Received by OWRD

Application for a Permit to Use SEP 0 2 2025

Groundwater

For Department Use: App. Number:

Salem, OR



Oregon Water Resources Department 725 Summer Street NE, Suite A Salem, Oregon 97301-1266 503-986-0900

Groundwater Application — Page 1

www.oregon.gov/OWRD

SECTION 1: APPLICANT INFORMATION AND SIGNATURE

Applicant				
NAME			0.5	PHONE (HM)
TA Operating LLC				
PHONE (WK) 440-808-7406	CE	ELL		FAX
MAILING ADDRESS 24601 Center Ridge Rd, Ste 200			П	
CITY Westlake	STATE OH	ZIP 44145	E-MAIL* KGelske@ta-petro.c	om
Organization				
NAME TA Operating LLC			PHONE 440-808-7406	FAX
MAILING ADDRESS 24601 Center Ridge Rd, Ste 200				CELL
CITY Westlake	STATE	ZIP 44145	E-MAIL* KGelske@ta-petro.co	m
Agant. The execution with extract to recover	ant the ann	licent in all	matters relating to this an	nlication
Agent – The agent is authorized to repres AGENT / BUSINESS NAME Andrey Chernishov / HBH Consulting	152 star		PHONE 503-644-6418	FAX FAX
MAILING ADDRESS 501 E First Street				CELL 503-951-0906
CITY Newberg	STATE	ZIP 97132	E-MAIL* achernishov@hbh-con	sulting.com
Note: Attach multiple copies as needed * By providing an e-mail address, consent is copies of the proposed and final order doc By my signature below I confirm that I am asking to use water specifica Evaluation of this application will I cannot use water legally until the Oregon law requires that a permin exempt. Acceptance of this application if I get a permit, I must not waste If development of the water use is the water use is the water use must be compatible. Even if the Department issues a powater to which they are entitled. I (we) affirm that the information of the water use is the water use is the water use is the water use and water to which they are entitled.	I understa Ily as descri be based or e Water Res t be issued ration does water. s not accord e with local bermit, I may	nd: bed in this an information sources Depute before beginner guarant ding to the transport to steel the comprehency have to steel in this app	application. In provided in the applicat artment issues a permit. Inning construction of any ee a permit will be issued. Berms of the permit, the permit and a permit will be so be sive land-use plans. In popusing water to allow se	proposed well, unless the use is ermit can be cancelled. nior water-right holders to get
Applicant Signature			Title if applicable	Date
Applicant Signature	Prin	t Name and	Title if applicable	Date

SECTION 2: PROPERTY OWNERSHIP

Please indicate if you own all the lands associated with the project from which the water conveyed, and used.	r is to be diverted,
YES, there are no encumbrances. YES, the land is encumbered by easements, rights of way, roads or other encumbran	ices.
NO, I have a recorded easement or written authorization permitting access. NO, I do not currently have written authorization or easement permitting access. NO, written authorization or an easement is not necessary, because the only affecte state-owned submersible lands, and this application is for irrigation and/or domestic NO, because water is to be diverted, conveyed, and/or used only on federal lands.	
Affected Landowners: List the names and mailing addresses of all owners of any lands t the applicant and that are crossed by the proposed ditch, canal or other work, even if the written authorization or an easement from the owner. (<i>Attach additional sheets if neces</i>)	e applicant has obtained ssary).
HPT TA Properties Trust 24601 Center Ridge Rd Westlake OH 44145	Received by OWRD

Salem, OR Legal Description: You must provide the legal description of: 1. The property from which the water is to be diverted, 2. Any property crossed by the proposed ditch, canal or other work, and 3. Any property on which the water is to be used as depicted on the map.

SECTION 3: WELL DEVELOPMENT

For

		IF LESS TH	AN 1 MILE:
WELL NO.	NAME OF NEAREST SURFACE WATER		
2	Onsite storm water pond & Tributary to Senecal Creek	N 500ft to Storm water Pond & N 1000ft to Tributary	~ 50ft
		×	
			3

Please provide any information for your existing or proposed well(s) that you believe may be helpful in evaluating your application. For existing wells, describe any previous alteration(s) or repair(s) not documented in the attached well log or other materials (attach additional sheets if necessary).

	Groundwater Application — Page 2
Department Use: App. Number:	Rev. 07/21

SECTION 3: WELL DEVELOPMENT, continued

Total maximum rate requested: 80 gpm (each well will be evaluated at the maximum rate unless you indicate well-specific rates and annual volumes in the table below).

The table below must be completed for each source to be evaluated or the application will be returned. If this is an existing well, the information may be found on the applicable well log. (<u>If a well log is available, please submit it in addition to completing the table</u>.) If this is a proposed well, or well-modification, consider consulting with a licensed well driller, geologist, or certified water right examiner to obtain the necessary information.

		=	1.2		37					PR	OPOSED U	SE	
OWNER'S WELL NAME OR NO.	PROPOSED	EXISTING	WELL ID (WELL TAG) NO.* OR WELL LOG ID**	FLOWING	CASING DIAMETER	CASING INTERVALS (IN FEET)	PERFORATED OR SCREENED INTERVALS (IN FEET)	SEAL INTERVALS (IN FEET)	MOST RECENT STATIC WATER LEVEL & DATE (IN FEET)	SOURCE AQUIFER***	TOTAL WELL DEPTH	WELL- SPECIFIC RATE (GPM)	ANNUAL VOLUME (ACRE-FEET)
2			102925		6" 0.250-Gauge	1-170	N/A	0-5 5-70 70-170	88 09/01/2010	Columbia River Basalt	221 ft	80	30.7
			-										
						7							
										2,2			

Licensed drillers are required to attach a Department-supplied Well Tag, with a unique Well ID or Well Tag Number to all new or newly altered wells. Landowners can request a Well ID for existing wells that do not have one. The Well ID is intended to serve as a unique identification number for each well.

For Department Use: App. Number:

SEP 0 2 2025
Salem, OR

Groundwater Application — Page 3

^{**} A well log ID (e.g. MARI 1234) is assigned by the Department to each log in the agency's well log database. A separate well log is required for each subsequent alteration of the well.

^{***} Source aguifer examples: Troutdale Formation, gravel and sand, alluvium, basalt, bedrock, etc.

SECTION 4: SENSITIVE, THREATENED OR ENDANGERED FISH SPECIES PUBLIC INTEREST INFORMATION

This information must be provided for your application to be accepted as complete. The Water Resources Department will determine whether the proposed use will impair or be detrimental to the public interest with regard to sensitive, threatened or endangered fish species if your proposed groundwater use is determined to have the potential for substantial interference with nearby surface waters.

To answer the following questions, use the map provided in <u>Attachment 3</u> or the link below to determine whether the proposed point of appropriation (POA) is located in an area where the Upper Columbia, the Lower Columbia, and/or the Statewide public interest rules apply.

For more detailed information, click on the following link and enter the TRSQQ or the Lat/Long of a POA and click on "Submit" to retrieve a report that will show which section, if any, of the rules apply: https://apps.wrd.state.or.us/apps/misc/lkp_trsqq_features/

If you need help to determine in which area the proposed POA is located, please call the customer service desk at (503) 986-0801.

Upper Columbia - OAR 690-033-0115 thru -0130	Received by OWRI
Is the well or proposed well located in an area where the Upper Columbia Rules apply?	SEP 0 2 2025
☐ Yes ■ No	Salem, OR

If yes, you are notified that the Water Resources Department will consult with numerous federal, state, local and tribal governmental entities so it may determine whether the proposed use is consistent with the "Columbia River Basin Fish and Wildlife Program" adopted by the Northwest Power Planning Council in 1994 for the protection and recovery of listed fish species. The application may be denied, heavily conditioned, or if appropriate, mitigation for impacts may be needed to obtain approval for the proposed use.

If yes, and if the Department determines that proposed groundwater use has the potential for substantial interference with nearby surface waters:

- I understand that the permit, if issued, will not allow use during the time period April 15 to September 30, except as provided in OAR 690-033-0140.
- I understand that the Department of Environmental Quality will review my application to determine if the proposed use complies with existing state and federal water quality standards.
- I understand that I will install and maintain water use measurement and recording devices as required by the Water Resources Department, and comply with recording and reporting permit condition requirements.

Lower Columbia - OAR 690-033-0220 thru -0230
Is the well or proposed well located in an area where the Lower Columbia rules apply?
Yes No
If yes, and the proposed groundwater use is determined to have the potential for substantial interference with nearby surface waters you are notified that the Water Resources Department will determine, by reviewing

For Department Use: App. Number:_

recovery plans, the Columbia River Basin Fish and Wildlife Program, and regional restoration programs applicable to threatened or endangered fish species, in coordination with state and federal agencies, as appropriate, whether the proposed use is detrimental to the protection or recovery of a threatened or endangered fish species and whether the use can be conditioned or mitigated to avoid the detriment.

If a permit is issued, it will likely contain conditions to ensure the water use complies with existing state and federal water quality standards; and water use measurement, recording and reporting required by the Water Resources Department. The application may be denied, or if appropriate, mitigation for impacts may be needed to obtain approval of the proposed use.

to obtain approval of the	proposed use.		
If yes, you will be require	d to provide the following information, if	applicable.	
	osed use is for more than one cubic foot pe 690, Division 86 (Water Management and 0) and is not subject to
If yes, provide a duse:	escription of the measures to be taken to a	assure reasonably eff	icient water
			Received by OWI
Statewide - OAR 690-033	<u>-0330 thru -0340</u>		
Is the well or proposed we	ell located in an area where the Statewide	rules apply?	SEP 0 2 2025
Yes No			Salem, OR
Department of Agriculture threatened and endanger If conditions cannot be ide		chieve "no loss of ess sential habitat of sen of essential T E fish h	ential habitat of nsitive (S) fish species." nabitat or no net loss of
USE	PERIOD OF USE	ANNUAL VO	DLUME (ACRE-FEET)
Commercial	Year-round		30.7
	9		
	ger e og r ag til		
Primary: Acres If you listed supplemental	acres, list the Permit or Certificate numbe	Nursery Use:	Acres rimary water right(s):
Indicate the maximum tot	tal number of acre-feet you expect to use i	n an irrigation seasor	1:

For Department Use: App. Number:_

- If the use is municipal or quasi-municipal, attach Form M
- If the use is domestic, indicate the number of households: (Exempt Uses: Please note that 15,000 gallons per day for single or group domestic purposes and 5,000 gallons per day for a single industrial or commercial purpose are exempt from permitting requirements.)
- If the use is **mining**, describe what is being mined and the method(s) of extraction (attach additional sheets if necessary):

SE	CTION 6: WATER MANAGEMENT	Received by OWRD
Α.	Diversion and Conveyance What equipment will you use to pump water from your well(s)?	SEP 0 2 2025
	Pump (give horsepower and type): 10 HP FPS / Franklin 100ST10D6B-0664 Other means (describe):	Salem, OR
	Provide a description of the proposed means of diversion, construction, and operation works and conveyance of water.	of the diversion
В.	Application Method What equipment and method of application will be used? (e.g., drip, wheel line, high-p (attach additional sheets if necessary)	ressure sprinkler)
С.	Conservation Please describe why the amount of water requested is needed and measures you prop waste; measure the amount of water diverted; prevent damage to aquatic life and ripa the discharge of contaminated water to a surface stream; prevent adverse impact to presurface waters (attach additional sheets if necessary). Flowmeter is installed for monitoring	rian habitat; prevent
SE	CTION 7: PROJECT SCHEDULE	
	 a) Date construction will begin: 08/17/2010 b) Date construction will be completed: 09/02/2010 c) Date beneficial water use will begin: Immediately 	
SE	CTION 8: RESOURCE PROTECTION	
act oe	granting permission to use water the state encourages, and in some instances requires, of ivities that may affect adjacent waterway or streamside area. See instruction guide for a rmit requirements from other agencies. Please indicate any of the practices you plan to the resources.	a list of possible
	Water quality will be protected by preventing erosion and run-off of waste or chemical Describe:	products.
	Excavation or clearing of banks will be kept to a minimum to protect riparian or stream Note: If disturbed area is greater than one acre, applicant should contact the Oregon D Environmental Quality to determine if a 1200C permit is required. Describe planned actions and additional permits required for project implementation:	

Other state and federal permits or contracts required and to be obtained, if a water right permit is granted:

List: N/A - well is already constructed and has active status with OHA (OR41 93593).

For Department Use: App. Number:____

SECTION 9: WITHIN A DISTRICT		
Check here if the point of appropriation irrigation or other water district.	on (POA) or place of use (POU) are	e located within or served by an
Irrigation District Name	Address	
irrigation district Name	Address	

SECTION 10: REMARKS

Use this space to clarify any information you have provided in the application (attach additional sheets if necessary).

Section 8 not applicable: well is already constructed and has active status with OHA (OR41 93593).

Attachments:

- 1. Legal Description
- 2. Well Log
- 3. Lease agreement
- 4. Map (24 x18")

Property shaded on map attachment for possible future landscaping use.

SEP 0 2 2025
Salem, OR

Minimum Requirements Checklist

Minimum Requirements (OAR 690-310-0040, OAR 690-310-0050 & ORS 537.140)

Include this checklist with the application

Check that each of the following items is included. The application will be returned if all required items are not included. If you have questions, please call the Water Rights Customer Service Group at (503) 986-0900.

Please submit the original application and signatures to the Water Resources Department. Applicants are

end	couraged to	o keep a copy of the completed applica	tion.
	SECTION SECTION SECTION	12: Property Ownership 13: Well Development	ature gered Fish Species Public Interest Information
	SECTION SECTION SECTION SECTION SECTION SECTION	Water Management Project Schedule Resource Protection Within a District	
Inc	lude the fo	ollowing additional items:	
	Land Use or signed		gnature of local planning department (must be an original)
	crossed by		from which the water is to be diverted, (2) any property ork, and (3) any property on which the water is to be used as
П		nount enclosed: \$	
_	See the De	epartment's Fee Schedule at www.oreg	gon.gov/owrd or call (503) 986-0900.
	Map th	hat includes the following items:	
		Permanent quality and drawn in ink	
		Even map scale not less than 4" = 1 m	ile (example: 1" = 400 ft, 1" = 1320 ft, etc.)
		North Directional Symbol	
		Township, Range, Section, Quarter/Qu	uarter, Tax
		Lots Reference corner on map	
		Location of each diversion, by referen (distances north/south and east/west	ce to a recognized public land survey corner
		Indicate the area of use by Quarter/Q	uarter and tax lot identified clearly.
		Number of acres per Quarter/Quarter irrigation, supplemental irrigation, or	r and hatching to indicate area of use if for primary nursery
		Location of main canals ditches nine	lines or flumes (if well is outside of the area of use)

Received by OWRD SEP 0 2 2025

Received by OWRD

Water-Use Permit Application Processing

SEP 0 2 2025

1. Completeness Determination

Salem, OR

The Department evaluates whether the application and accompanying map contain all of the information required under OAR 690-310-0040 and OAR 690-310-0050. The Department also determines whether the proposed use is prohibited by statute. If the Department determines that the application is incomplete, all fees have not been paid, or the use is prohibited by statute, the application and all fees submitted are returned to the applicant.

2. Initial Review

The Department reviews the application to determine whether water is available during the period requested, whether the proposed use is restricted or limited by rule or statute, and whether other issues may preclude approval of or restrict the proposed use. An Initial Review (IR) containing preliminary determinations is mailed to the applicant. The applicant has 14 days from the mailing date to withdraw the application from further processing and receive a refund of all fees paid minus \$310. The applicant may put the application on hold for up to 180 days and may request additional time if necessary.

3. Public Notice

Within 7 days of the mailing of the initial review, the Department gives <u>public notice</u> of the application in the weekly notice published by the Department at <u>www.oregon.gov/owrd</u>. The public comment period is 30 days from publication in the weekly notice.

4. Proposed Final Order Issued

The Department reviews any comments received, including comments from other state agencies related to the protection of sensitive, threatened or endangered fish species. Within 60 days of completion of the IR, the Department issues a Proposed Final Order (PFO) explaining the proposed decision to deny or approve the application. A PFO proposing approval of an application will include a draft permit, and may request additional information or outstanding fees required prior to permit issuance.

5. Public Notice

Within 7 days of issuing the PFO, the Department gives public notice in the weekly notice. Notice includes information about the application and the PFO. Protest must be received by the Department within 45 days after publication of the PFO in the weekly notice. Anyone may file a protest. The protest filing fee is \$480.00 for the applicant and \$950.00 for non-applicants. Protests are filed on approximately 10 percent of Proposed Final Orders. If a protest is filed the Department will attempt to settle the protest but will schedule a contested case hearing if necessary.

6. Final Order Issued

If no protests are filed, the Department can issue a Final Order within 60 days of the close of the period for receiving protest. If the application is approved, a permit is issued. The permit specifies the details of the authorized use and any terms, limitations or conditions that the Department deems appropriate

Land Use SEP 0 2 2025 Information Form Salem, OR



Oregon Water Resources Department 725 Summer Street NE, Suite A Salem, Oregon 97301-1266 (503) 986-0900 www.oregon.gov/OWRD

NOTE TO APPLICANTS

In order for your application to be processed by the Oregon Water Resources Department (OWRD), this Land Use Information Form must be completed by a local government planning official in the jurisdiction(s) where your water right will be diverted, conveyed, used, and developed. The planning official may choose to complete the form while you wait or return the "Receipt Acknowledging Request for Land Use Information" to you. Applications received by OWRD without the Land Use Information Form, or the signed receipt, will be returned to you. **IMPORTANT:** Please note that while OWRD can accept a signed receipt as part of intake for an application for a new permit to use or store water, a completed Land Use Information Form is required for OWRD's acceptance of all other applications. Please be aware that your application cannot be approved without land use approval.

This form is NOT required if:

- 1) Water is to be diverted, conveyed, and used on federal lands only; OR
- 2) The application is for a water right transfer, allocation of conserved water, exchange, permit amendment, or ground water registration modification, and <u>all</u> of the following apply:
 - **a.** The existing and proposed water use is located entirely within lands zoned for exclusive farm-use or within an irrigation district;
 - b. The application involves a change in place of use only;
 - **c.** The change does not involve the placement or modification of structures, including but not limited to water diversion, impoundment, distribution facilities, water wells and well houses; **and**
 - d. The application involves irrigation water uses only.

NOTE TO LOCAL GOVERNMENTS

The person presenting the attached Land Use Information Form is applying for a new water right or modifying an existing water right. The Oregon Water Resources Department (OWRD) requires applicants to obtain land use information to ensure the water right does not result in land uses that are incompatible with your comprehensive plan. Please complete the form and return it to the applicant for inclusion in their application. **NOTE:** For new water right applications only, if you are unable to complete this form while the applicant waits, you may complete the "Receipt Acknowledging Request for Land Use Information" and return it to the applicant.

You will receive notice via OWRD's weekly Public Notice once the applicant formally submits their request to OWRD. The notice will give more information about OWRD's water right process and provide additional comment opportunities. If you previously only completed the receipt for an application for a new permit to use or store water, you will have 30 days from the Public Notice date to complete the Land Use Information Form and return it to OWRD. Your attention to this request for information is greatly appreciated. If you have questions concerning this form, please contact OWRD's Customer Service Group at 503-986-0900 or WRD_DL_customerservice@water.oregon.gov.

SEP 0 2 2025
Salem, OR

Last Revised: 10/2023

This page intentionally left blank.

Land Use Information Form

Received by OWBREGON

SEP 0 2 2025



Salem, OR

Oregon Water Resources Department 725 Summer Street NE, Suite A Salem, Oregon 97301-1266 (503) 986-0900 www.oregon.gov/OWRD

							1	12.16.2.1		
NAME TA Opera	ting LLC							PHO 440-	NE 808-7406	
	ADDRESS nter Ridge		11							
CITY				STATE	ZIP	EMAIL				
Westlake	5	y *5		ОН	44145	KGelsk	e@ta-petro.	com		
(transporte	ude the fo	ollowing in	developed	. Applicants	for municip	oal use, or	irrigation us	taken from i ses within irri equested be	igation di	
Township	Range	Section	<i>1</i> / ₄ 1/ ₄	Tax Lot	Plan Designat Rural Residen			Water to be:		Proposed Land Use:
45	1W	9	NE ¼ SE ¼	1100	Indust		Diverted	Conveyed	Used	Commercial
							Diverted	Conveyed	Used	
							Diverted	Conveyed	Used	
							Diverted	Conveyed	Used	
List all cou	nties and	cities wher	e water is	proposed t	o be diverte	ed. conve	ved_and/or u	used or devel	oned:	
Marion C		-Tes-		р.оросси		, , , , , , , ,	,,,		opea.	
		ond Use Info		Form must I	be complete	ed and sul	omitted for <u>e</u>	each county a	and city, a	as applicable.
Permit		Store Water	□w	regon Wate ater Right Tra change of Wa	ansfer [Permit			er Registra	ation Modification
Source of v	vater:	Reservoi	r/Pond	Ground	Water	Surfac	e Water (nam	e)		
Estimated	quantity o	of water ne	eded:	80	cubic fee	et per seco	nd g al	lons per minut	te 🗌	acre-feet
Intended u	se of wat		rrigation Municipal		nmercial si-Municipal		ustrial ream		: for	_household(s)
Briefly des									-	
Water su	pply well	for Trave	l Centers	of America	a facility					

Note to applicant: For new water right applications only, if the Land Use Information Form cannot be completed while you wait, please have a local government representative sign the receipt on the bottom of page 4 and include it with the application filed with the Oregon Water Resources Department.

See Page 4 →

For Local Government Use Only

The following section must be completed by a planning official from each county and city in the project will be located entirely within the city limits. In that case, only the city planning agency must complete this form. This deals only with the local land use plan. Do not include approval for activities such as building or grading permits.

Please check the appropriate box	below and provide the requested info	ormation		
	d water use(s), including proposed construction. Cite applicable ordinance section(s): MCC			
approvals as listed in the table below. already been obtained. Record of Action	d water use(s), including proposed construction (Please attach documentation of applicable labon/land-use decision and accompanying findir have not ended, check "Being Pursued."	nd-use approv	als which have	
Type of Land-Use Approval Needed (e.g., plan amendments, rezones, conditional-use permits, etc.)	Cite Most Significant, Applicable Plan Policies & Ordinance Section References	Land-	Use Approval:	
		Obtained Denied	☐ Being Pursued ☐ Not Being Pursued	
		☐ Obtained ☐ Denied	☐ Being Pursued ☐ Not Being Pursued	
		☐ Obtained ☐ Denied	☐ Being Pursued ☐ Not Being Pursued	
		☐ Obtained ☐ Denied	☐ Being Pursued ☐ Not Being Pursued	
Name: Gillian Peden	Title:Assist	_Title:Assistant Planner		
Signature: <i>GP</i>	Date:	2025	-	
Governmental Entity: Marion County	Phone: <u>503-5</u>	588-5038		
Receipt Ackr	nowledging Request for Land Use Info	rmation		
this form while the applicant waits, you ma have 30 days from the date of OWRD's Pub Oregon Water Resources Department. Plea	the applicant. For new water right applications by complete this receipt and return it to the apploic Notice of the application to submit the compase note while OWRD can accept a signed receip completed Land Use Information Form is required	licant. If you sign eleted Land Use t as part of intal	n the receipt, you will Information Form to se for an application	
	Title:			
Governmental Entity:	Phone:			

APPENDICES

- 1. Legal Description
- 2. Well Log
- 3. Lease Agreement
- 4. Map (24 x 18")

SEP 0 2 2025
Salem, OR

	- 4 ² N N		Reel 2777	Page
SS 7	Stewart Title Guaranty Fo	received for recei	nat the within is ording on lock e No/instrument/m, Records of	instrument was M., and recorded in on page icrofilm/reception
577	SPECIAL WARI (Corporate			Portland
AmeriTitle	TA Operating LLC, a limited liability company duly organic Delaware, Grantor, conveys and warrants to HPT TA Properties Trust, a Maryland real estate investme free of encumbrances, except as specifically set forth there. See legal description attached hereto as Exhibit A.	ent trust Grantee the fol	lowing de	scribed real propert
		E)		z
	(IF SPACE INSUFFICIENT, CONTINUE	DESCRIPTION ON REVERSE S	IDE)	
	Grantor hereby covenants with said Grantee that the grant- Grantor has good right and lawful authority to sell and conv title to said land and will defend the same against the lawful Grantor.	ev said land: that the qu	antor here	eby fully warrants th
	The true and actual consideration paid for this transfer, (Here, comply with the requirements of ORS 93.030.)	stated in terms of dollar	s, is \$ 10.	00

Received by OWRD
SEP 0 2 2025
Salem, OR

Done by order of the grantor's board of directors with its corporate seal, if any affixed on January 31, 2007.

By: John R. Hoadley
Its: Executive Vice President and Treasurer

THIS INSTRUMENT WILL NOT ALLOW USE OF THE PROPERTY DESCRIBED IN THIS INSTRUMENT IN VIOLATION OF APPLICABLE LAND USE LAWS AND REGULATIONS. BEFORE SIGNING OR ACCEPTING THIS INSTRUMENT, THE PERSON ACQUIRING FEE TITLE TO THE PROPERTY SHOULD CHECK WITH THE APPROPRIATE CITY OR COUNTY PLANNING DEPARTMENT TO VERIFY APPROVED USES AND TO DETERMINE ANY LIMITS ON LAWSUITS AGAINST FARMING OR FOREST PRACTICES AS DEFINED IN ORS 30.930



SEP 0 2 2025
Salem, OR

Exhibit A

Legal Description

[See copy attached.]

SEP 0 2 2025
Salem, OR

3.056 Portland, OR
21856 Vents Road, NE
Aurora, OR 97002
(TCA Site No. 56 - Portland)

Exhibit "A "

A parcel of land located in the Amable Arquait Donation Land Claim No. 45 in the Southeast one-quarter of Section 9, Township 4 South, Range 1 West of the Willamette Meridian, Marion County, Oregon and more particularly described as follows:

Beginning at an iron rod on the Easterly right-of-way line of County Road No. 425, said point being South 08°07'00" East along the centerline of said County Road a distance of 1,666.58 feet (the deed calls this 1,667.50 feet) and North 73°21'00" East a distance of 30.34 feet from the Southeast corner of the David Crawford Donation Land Claim No. 41; thence continuing North 73°21'00" East a distance of 629.64 feet (the deed calls this 630.05 feet); thence North 30°08'00" East a distance of 721.88 feet (the deed calls this 722.11 feet) to a point on the Southerly line of that parcel of land conveyed to Louis N. Racette et al, and recorded in Volume269, page 508, Marion County Deed Records; thence North 73°21'00" East along the Southerly line of said Racette property a distance of 479.69 feet (the deed calls this 479.64 feet) to a point on the Westerly right-of-way line of Pacific Highway (Interstate No. 5); thence South 30°08'00" West along said Westerly right-of-way line a distance of 1897.70 feet (the deed calls this 1898.17 feet); thence South 42°36'38" West, (the deed calls this South 42°36' West) along said Westerly right-of-way line a distance of 189.54 feet (the deed calls this 189.59 feet); thence South 46°20'55" West (the deed calls this South 46°19'00" West) along said Westerly right-of-way line a distance of 161.22 feet (the deed calls this 161.43 feet); thence North 59°54'00" West along said right-of-way line a distance of 104.21 feet (the deed calls this 104.28 feet); thence continuing along said right-of-way line Northwesterly along a 246.48 foot radius curve to the right, through a central angle of 00°40'02" an arc distance of 2.87 feet (the long chord of sald curve bears North 08°29'44" West (the deed calls this North 08°28' West a distance of 2.87 feet); thence South 81°53'00" West along said right-of-way line a distance of 3.65 feet; thence North 59°54'00" West along said right-of-way line a distance of 7.17 feet to a point on the Easterly right-of-way line of County Road No. 425; thence North 08°07'00" West along said Easterly right-of-way line a distance of 900.66 feet (the deed calls this 900.92 feet) to the point of beginning.

Received by OWRD
SEP 0 2 2025
Salem. OR

REEL:2777

PAGE: 199

February 23, 2007, 03:26 pm.

CONTROL #: 189449

State of Oregon County of Marion Received by OWRD SEP 0 2 2025

Salem, OR

I hereby certify that the attached instrument was received and duly recorded by me in Marion County records:

FEE: \$ 41.00

BILL BURGESS COUNTY CLERK

THIS IS NOT AN INVOICE.

Received by OWRD

SEP 0 2 2025

STATE OF OREGON WATER SUPPLY WELL REPORT (as required by ORS 537.765 & OAR 690-205-0210) 5EP 24 2010

WATER RESOURCES DEPT SALEM, OREGON

Salem, OR WELL LABEL # L 102925 START CARD# 205285

(1) LAND OWNER Owner Well I.D.	(9) LOCATION OF WELL (legal description	1)	
First Name Last Name	County MARION Twp 4 S N/S Range		
Company TA OPERATING LLC	Sec 9 1/4 of the 1/4 Tax I	ot 1100	
Address 24601 CENTER RIDGE RD	Tax Map Number Lot	DV6 DD	
City WESTLAKE State OH Zip 97493	Lat " or 45.23666667	DIMS OF DD	
(2) TYPE OF WORK New Well Deepening Conversion	Long ° " or 122.80694444	DMS or DD	
Alteration (repair/recondition) Abandonment	Street address of well Nearest address		
	21856 BENTS RD NE AURORA, OR		
(3) DRILL METHOD			
Rotary Air Rotary Mud Cable Auger Cable Mud	(10) STATIC WATER LEVEL Date SWL(ps		
Reverse Rotary Other		i) + SWL(fi)	
(4) PROPOSED USE Domestic Irrigation Community	Existing Well / Predeepening		
Industrial/ Commercial Livestock Dewatering	Completed Well 09-01-2010	88	
Thermal Injection Other	Flowing Artesian? Dry Hole		
	WATER BEARING ZONES Depth water was first		
(5) BORE HOLE CONSTRUCTION Special Standard Attach copy)			
Depth of Completed Well 221 ft.	08-18-2010 28 28 5 08-23-2010 72 150 30	15 70	
BORE HOLE SEAL sacks/ Dia From To Material From To Amt Ibs	09-01-2010 170 221 125	88	
14 0 20 Bentonite 0 5 6 S	0301-2010 170 221 123		
12 20 70 Cement 5 70 40 S			
10 70 170 Cement 70 170 28 S	(14) WELL VOO		
6 170 221	(11) WELL LOG Ground Elevation		
How was seal placed: Method A B XC XD E	Material Fro	om To	
Other	Black top over crushed rock	0 1	
Backfill placed from ft. to ft. Material	Silty brown clay with trace of sand	28	
Filter pack from fi. to fi. Material Size		28 36 36 54	
Explosives used: Yes Type Amount		54 56	
(6) CASING/LINER		56 72	
Casing Liner Dia + From To Gauge Stl Plstc Wid Thrd		72 85	
○	Red very fine sand and gravel	85 96	
	Brown very sandy soft clay w/fine sandy seams	96 126	
		126 144	
		144 150	
		150 170 170 176	
Shoe Inside Outside Other Location of shoe(s)		176 185	
Temp casing X Yes Dia From To	The state of the s	185 188	
(7) PERFORATIONS/SCREENS	Semi-fractured gray basalt	188 190	
Perforations Method	Hard dark gray basalt with some fractured seams	190 221	
Screens Type Material			
Perf/S Casing/ Screen Scrn/slot Slot # of Tele/ creen Liner Dia From To width length slots pipe size	Date Started 08-17-2010 Completed 09-	02-2010	
Tom to most longer story	(unbonded) Water Well Constructor Certification		
	I certify that the work I performed on the construction,	deepening, alteration, or	
	abandonment of this well is in compliance with Ore		
	construction standards. Materials used and information re	eported above are true to	
	the best of my knowledge and belief.		
(8) WELL TESTS: Minimum testing time is 1 hour	License Number 1629 Date 09-07-	-2010	
Pump Bailer • Air Flowing Artesian	Password : (if filing electronically)		
Yield gal/min Drawdown Drill stem/Pump depth Duration (hr)	Signed		
125 220 3	(bonded) Water Well Constructor Certification		
	I accept responsibility for the construction, deepening, a		
The state of the s	work performed on this well during the construction dates		
Temperature 53 / °F Lab analysis Yes By	performed during this time is in compliance with Ord construction standards. This report is true to the best of my		
Water quality concerns? Yes (describe below) From To Description Amount Units	The state of the s		
	License Number) 1273 Date 09-07-2010 Password: (if frigg electromically)		
	Signed Flora State		
	Contact Info (optional)		

ORIGINAL - WATER RESOURCES DEPARTMENT
THIS REPORT MUST BE SUBMITTED TO THE WATER RESOURCES DEPARTMENT WITHIN 30 DAYS OF COMPLETION OF WORK

6-17267

Received by OWRD SEP 0 2 2025
Salem, OR

THIRD AMENDED AND RESTATED LEASE AGREEMENT NO. 2,

dated as of May 15, 2023

by and among

HPT TA PROPERTIES TRUST and HPT TA PROPERTIES LLC,

AS LANDLORD,

and

TA OPERATING LLC,

AS TENANT

SEP 0 2 2025

Salem, OR

TABLE OF CONTENTS

	I	Page
ARTICLE 1	DEFINITIONS	1
1.1	"AAA"	
1.2	"Additional Charges"	
1.3	"Affiliated Person"	
1.4	"Agreement"	
1.5	"Appellate Rules"	
1.6	"Applicable Laws"	
1.7	"Arbitration Award"	
1.8	"Award"	
1.9	"BP Affiliate"	
1.10	"BP Parent"	
1.11	"Business Day"	
1.12	"Capital Expenditure"	
1.13	"Capital Addition"	3
1.14	"Claims"	
1.15	"Code"	
1.16	"Condemnation"	
1.17	"Condemnor"	
1.18	"Control"	
1.19	"Declaration"	
1.20	"Default"	
1.21	"Disputes"	
1.22	"Distribution"	
1.23	"Easement Agreement"	
1.24	"Effective Date"	
1.25	"Encumbrance"	
1.26	"Entity"	
1.27	"Environment"	
1.28	"Environmental Notice"	
1.29	"Environmental Obligation"	
1.30	"Environmental Report"	
1.31	"Event of Default"	
1.32	"Extended Term"	
1.33	"Financial Officer's Certificate"	
1.34	"Fixed Term"	
1.35	"Fixtures"	
1.36	"GAAP"	
1.37	"Government Agencies"	
1.38	"Ground Leases"	
1.39	"Guarantor"	
1.40	"Guaranty"	
1.41	"Hazardous Substances"	
1.42	"IFRS"	

Received by OWRD

SEP 0 2 2025

"Impositions"	Salem,	OF
"Insurance Requirements"	***************************************	6
'Interest Rate'		
"Land"		
"Landlord"		
"Landlord Default"		
"Landlord Liens"		
Lease Year"		
<u>Lease Year</u> "Leased Improvements"		
"Leased Property"		
"Legal Requirements"		
" <u>Lien</u> "		
"Manager"		
"Merger"		
Minerals"		
"Minimum Net Worth"		
'Minimum Rent''	•••••	8
'Minimum Rent Reduction Limit''	•••••	8
'Net Worth''		8
'Non-Recourse Parties''		
'Notice"		
<u>'Offer</u> "		
'Offer Notice"		
'Offered Property"		
'Officer's Certificate'		9
'Operating Rights"		9
'Other Leases"		
'Overdue Rate"		
'Parent"		
Percentage Reduction"		
'Permitted Encumbrances''		
'Permitted Use"		
"Person"		
'Prior Lease"		
'Property"		
Property Mortgage"		9
'Property Mortgagee"		
'Qualifying Guarantor"		
Real Property"		10
'Related Person"		
'Rent"		
"Rules"		
" <u>SARA</u> "		10
" <u>SEC"</u>		10
"Shell"		10
"Shell Agreement"		

Received by OWRD SEP 0 2 2025

Salem, OR

1.89	"Shell SNDA"	11
1.90	"State"	11
1.91	"Successor Landlord"	11
1.92	"Superior Landlord"	11
1.93	"Superior Lease"	
1.94	"Superior Mortgage"	
1.95 1.96	"Superior Mortgagee"	11
1.97	"TCA"	
1.98	"Tenant"	
1.99	"Tenant's Personal Property"	
1.100	"Term"	
1.101	"Total Assets"	
1.102	"Total Liabilities"	
1.103		
1.104	"Unsuitable for Its Permitted Use"	12
ARTICLE 2	LEASED PROPERTY AND TERM	12
2.1	Leased Property	
2.2	Condition of Leased Property	
2.3	Term	
2.4	Extended Terms	13
2.5	Right to Repool Properties	
2.6	Right of First Offer	
ARTICLE 3	RENT	15
3.1	Rent	15
3.2	Late Payment of Rent, Etc	
3.3	Net Lease, Etc	
3.4	No Termination, Abatement, Etc	
3.5	Prepayment	
ARTICLE 4	USE OF THE LEASED PROPERTY	18
4.1	Permitted Use	
4.2	Environmental Matters	
4.3	Ground Leases	
4.4	Shell Agreement	22
ARTICLE 5	MAINTENANCE AND REPAIRS	23
5.1	Maintenance and Repair	23
5.2	Tenant's Personal Property	
5.3	Yield Up	24
ARTICLE 6	IMPROVEMENTS, ETC.	25
ARTICLE 7	LIENS	
ARTICLE 8	PERMITTED CONTESTS	
ARTICLE 9	INSURANCE AND INDEMNIFICATION	
9.1	General Insurance Requirements	26

Received by OWRD SEP 0 2 2025

	Salem, OR	\$
9.2	Waiver of Subrogation	26
9.3	Form Satisfactory, Etc.	26
9.4	<u>Self-Insurance</u>	
9.5	Indemnification of Landlord	27
ARTICLE 10	CASUALTY	28
10.1	Insurance Proceeds	
10.2	Damage or Destruction	
10.3	Damage Near End of Term	
10.4	Restoration of Tenant's Personal Property	
10.5	No Abatement of Rent	
10.6	<u>Waiver</u>	29
ARTICLE 11	CONDEMNATION	29
11.1	Total Condemnation, Etc	
11.2	Partial Condemnation	
11.3	Abatement of Rent	
11.4	Temporary Condemnation	
11.5	Allocation of Award	
ARTICLE 12	DEFAULTS AND REMEDIES	31
12.1	Events of Default	
12.2	Remedies	
12.3	Tenant's Waiver	
12.4	Application of Funds.	
12.5	Landlord's Right to Cure Tenant's Default	
ARTICLE 13	HOLDING OVER	
ARTICLE 14	LANDLORD DEFAULT	
ARTICLE 15	PURCHASE OF TENANT'S PERSONAL PROPERTY	36
ARTICLE 16	SUBLETTING AND ASSIGNMENT	
16.1	Subletting and Assignment	
16.2	Required Sublease Provisions	
16.3	Permitted Subleases and Assignments	
16.4	Sublease Limitation	39
ARTICLE 17	ESTOPPEL CERTIFICATES AND OPERATING STATEMENTS	
17.1	Estoppel Certificates	
17.2	Operating Statement	39
ARTICLE 18	LANDLORD'S RIGHT TO INSPECT	40
ARTICLE 19	EASEMENTS; ZONING	40
19.1	Grant of Easements	40
19.2	Exercise of Rights by Tenant	40
19.3	Permitted Encumbrances	
19.4	Zoning	41
ARTICLE 20	PROPERTY MORTGAGES	41
	Landlord May Crant Liens	41

Received by OWRD SEP 0 2 2025

Salem, OR

	20.2	Subordination of Lease	41
	20.3	Notice to Mortgagee and Superior Landlord	
ARTI	CLE 21	ADDITIONAL COVENANTS OF LANDLORD AND TENANT	43
	21.1	Prompt Payment of Indebtedness	
	21.2	Maintenance of Accounts and Records	
	21.3	Notice of Litigation, Etc.	
	21.4	Distributions, Payments to Affiliated Persons, Etc	
	21.5	Merger; Sale of Assets	
	21.6	REIT Qualification	
ARTI	CLE 22	ARBITRATION	44
	22.1	Disputes	
	22.2	Selection of Arbitrators	
	22.3	Location of Arbitration	
	22.4	Scope of Discovery	
	22.5	Arbitration Award	
	22.6	Appeals	
	22.7	Final Judgment	
	22.8	Intended Beneficiaries	
ARTI	CLE 23	REPRESENTATIONS	
Aucii	23.1	Landlord's Representations	
	23.2	Tenant's Representations	
ADTI	CLE 24	MISCELLANEOUS	
AKII	24.1	Limitation on Payment of Rent	
	24.2	No Waiver	
	24.2	Remedies Cumulative	
	24.4	Severability	
	24.5	Acceptance of Surrender	
	24.6	No Merger of Title	
	24.7	Conveyance by Landlord	
	24.8	Quiet Enjoyment	
	24.9	No Recordation	
	24.10	Notices	
	24.11		
	24.12		
	24.13	Applicable Law, Etc.	
		Right to Make Agreement	
		Confidentiality	
	24.16		
	24.17	Exculpation Nonliability of Trustees	51

THIRD AMENDED AND RESTATED LEASE AGREEMENT NO. 2

THIS THIRD AMENDED AND RESTATED LEASE AGREEMENT NO. 2 is dated as of May 15, 2023 (the "Effective Date"), by and among HPT TA PROPERTIES TRUST, a Maryland real estate investment trust, and HPT TA PROPERTIES LLC, a Maryland limited liability company (collectively, "Landlord"), and TA OPERATING LLC, a Delaware limited liability company ("Tenant").

WITNESSETH:

WHEREAS, Landlord and Tenant are parties to that certain Second Amended and Restated Lease Agreement No. 2, dated as of October 14, 2019, as amended from time to time (as so amended, the "Prior Lease");

WHEREAS, Tenant is a subsidiary of TravelCenters of America Inc., a Maryland corporation ("TCA") and TCA has entered into that certain Agreement and Plan of Merger, dated as of February 14, 2023, with BP Products North America Inc. ("Parent") and an indirect wholly-owned subsidiary of Parent ("Merger Subsidiary"), pursuant to which, among other things, Merger Subsidiary will merge with and into TCA (the "Merger"), in accordance with the applicable provisions of the Maryland General Corporation Law, on the terms and subject to the conditions set forth therein; and

WHEREAS, in connection with the Merger, Landlord and Tenant wish to amend and restate the Prior Lease as herein provided;

NOW, THEREFORE, in consideration of the mutual covenants herein contained and other good and valuable consideration, the mutual receipt and legal sufficiency of which are hereby acknowledged, Landlord and Tenant hereby agree that, from and after the Effective Date, the Prior Lease is hereby amended and restated in its entirety as follows:

ARTICLE 1 DEFINITIONS

For all purposes of this Agreement, except as otherwise expressly provided or unless the context otherwise requires, (a) the terms defined in this Article shall have the meanings assigned to them in this Article and include the plural as well as the singular, (b) all accounting terms not otherwise defined herein shall have the meanings assigned to them in accordance with GAAP, (c) all references in this Agreement to designated "Articles", "Sections" and other subdivisions are to the designated Articles, Sections and other subdivisions of this Agreement, and (d) the words "herein", "hereof", "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision.

- 1.1 "AAA" shall have the meaning given such term in Section 22.1.
- 1.2 "Additional Charges" shall have the meaning given such term in Section 3.1.2.
- 1.3 "<u>Affiliated Person</u>" shall mean, with respect to any Person, (a) in the case of any such Person which is a partnership, limited liability company, corporation or other Entity,

Salem, OR

any owner of a partnership interest, limited liability company interest, share of capital stock or other equity interest in that Entity, (b) any other Person which is a parent, a subsidiary, or a subsidiary of a parent with respect to such Person or to one or more of the Persons referred to in the preceding clause (a), (c) any other Person who is an officer, director, trustee or employee of, or partner in or member of, such Person or any Person referred to in the preceding clauses (a) and (b), and (d) any other Person who is a member of the immediate family of such Person or of any Person referred to in the preceding clauses (a) through (c).

- 1.4 "<u>Agreement</u>" shall mean this Third Amended and Restated Lease Agreement No. 2, including all exhibits attached hereto, as it and they may be amended from time to time as herein provided.
 - 1.5 "Appellate Rules" shall have the meaning given such term in Section 22.7.
- 1.6 "Applicable Laws" shall mean all applicable laws, statutes, regulations, rules, ordinances, codes, licenses, permits, notices and orders, from time to time in existence, of all courts of competent jurisdiction and Government Agencies, and all applicable judicial and administrative and regulatory decrees, judgments and orders, including common law rulings and determinations, relating to injury to, conservation of, or the protection of, real or personal property, or human health or the Environment, including, without limitation, all valid and lawful requirements of courts and other Government Agencies pertaining to reporting, licensing, permitting, investigation, remediation and removal of underground improvements (including, without limitation, treatment or storage tanks, or water, natural gas or oil wells), or emissions, discharges, releases or threatened releases of Hazardous Substances, chemical substances, pesticides, petroleum or petroleum products, pollutants, contaminants or hazardous or toxic substances, materials or wastes whether solid, liquid or gaseous in nature, into the Environment, or relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Substances, underground improvements (including, without limitation, treatment or storage tanks, or water, gas or oil wells), or pollutants, contaminants or hazardous or toxic substances, materials or wastes, whether solid, liquid or gaseous in nature.
 - 1.7 "Arbitration Award" shall have the meaning given such term in Section 22.5.
- 1.8 "Award" shall mean all compensation, sums or other value awarded, paid or received by virtue of a total or partial Condemnation of any Property (after deduction of all reasonable legal fees and other reasonable costs and expenses, including, without limitation, expert witness fees, incurred by Landlord, in connection with obtaining any such award).
- 1.9 "BP Affiliate" shall mean an Entity which is domiciled in the United States and is an Affiliated Person with respect to BP Parent.
- 1.10 "BP Parent" shall mean BP p.l.c., a public limited company incorporated under the laws of England and Wales, together with its successors by merger, consolidation or transfer of all or substantially all of its assets.
- 1.11 "Business Day" shall mean any day other than Saturday, Sunday, or any other day on which banking institutions in The Commonwealth of Massachusetts are authorized by law or executive action to close.

- 1.12 "<u>Capital Expenditure</u>" shall mean any expenditure treated as capital in nature in ance with GAAP. accordance with GAAP.
- "Capital Addition" shall mean, with respect to any Property, any renovation. repair or improvement to such Property, including without limitation any item the expense of which is a Capital Expenditure.
 - "Claims" shall have the meaning given such term in Article 8.
- "Code" shall mean the Internal Revenue Code of 1986 and, to the extent applicable, the Treasury Regulations promulgated thereunder, each as from time to time amended.
- 1.16 "Condemnation" shall mean, with respect to any Property, or any portion thereof, (a) the exercise of any governmental power with respect to such Property, whether by legal proceedings or otherwise, by a Condemnor of its power of condemnation, (b) a voluntary sale or transfer of such Property by Landlord to any Condemnor, either under threat of condemnation or while legal proceedings for condemnation are pending, or (c) a taking or voluntary conveyance of such Property, or any interest therein, or right accruing thereto or use thereof, as the result or in settlement of any condemnation or other eminent domain proceeding affecting such Property, whether the same shall have actually been commenced or only threatened by the relevant Condemnor.
- 1.17 "Condemnor" shall mean any public or quasi-public Person, having the power of Condemnation.
- "Control" shall mean, with respect to any specified Person, the possession, directly or indirectly, of the power to direct, without the consent of any other Person required, or to cause the direction of the management or policies of such Person, whether through ownership of voting securities or other ownership interests, by contract or otherwise, provided, that, "Control" shall not be deemed absent solely because another Person shall have customary liquidity rights and/or veto power with respect to major decisions. "Controlled," "Controlling" and "Controlled by" shall have correlative meanings.
 - "**Declaration**" shall have the meaning given such term in Section 24.18. 1.19
- 1.20 "Default" shall mean any event or condition which with the giving of notice and/or lapse of time would be an Event of Default.
 - "Disputes" shall have the meaning given such term in Section 22.1. 1.21
- "Distribution" shall mean (a) any declaration or payment of any dividend (except 1.22 ordinary cash dividends payable in common stock or other equity interests of Tenant) on or in respect of any shares of any class of capital stock or other equity interests of Tenant, (b) any purchase, redemption, retirement or other acquisition by Tenant of any shares of any class of capital stock or other equity interests of Tenant, (c) any other distribution on or in respect of any shares of any class of capital stock or other equity interests of Tenant or (d) any return of capital to shareholders or other equity interest holders of Tenant.

Salem, OR

- 1.23 "Easement Agreement" shall mean any conditions, covenants and restrictions, easements, declarations, licenses and other agreements which are Permitted Encumbrances and such other agreements as may be granted in accordance with Section 19.1.
- 1.24 "Effective Date" shall have the meaning given such term in the preamble to this Agreement.
 - 1.25 "Encumbrance" shall have the meaning given such term in Section 20.1.
- 1.26 "Entity" shall mean any corporation, general or limited partnership, limited liability company or partnership, stock company or association, joint venture, association, company, trust, bank, trust company, land trust, business trust, real estate investment trust, cooperative, any government or agency, authority or political subdivision thereof or any other entity.
- 1.27 "Environment" shall mean soil, surface waters, ground waters, land, biota, sediments, surface or subsurface strata and ambient air.
- 1.28 "Environmental Notice" shall have the meaning given such term in Section 4.3.1.
- 1.29 "Environmental Obligation" shall have the meaning given such term in Section 4.3.1.
- 1.30 "Environmental Report" shall have the meaning given such term in Section 4.3.2.
 - 1.31 "Event of Default" shall have the meaning given such term in Section 12.1.
 - 1.32 "Extended Term" shall have the meaning given such term in Section 2.4.
- 1.33 "Financial Officer's Certificate" shall mean a certificate of the chief executive officer, chief financial officer or chief accounting officer (or such officers' authorized designee) of such Person, duly authorized, accompanying the statements required to be delivered by such Person pursuant to Section 17.2, in which such officer shall certify that such statements have been prepared based on the accounting records and systems used by Tenant in the ordinary course of its business.
 - 1.34 "Fixed Term" shall have the meaning given such term in Section 2.3.
 - 1.35 "Fixtures" shall have the meaning given such term in Section 2.1(d).
- 1.36 "GAAP" shall mean generally accepted accounting principles consistently applied, it being understood that if Tenant or Guarantor or another relevant Person generally observes IFRS in its financial accounting then references to GAAP shall mean the IFRS.
- 1.37 "Government Agencies" shall mean any court, agency, authority, board (including, without limitation, environmental protection, planning and zoning), bureau,

commission, department, office or instrumentality of any nature whatsoever of any governmental, or quasi-governmental unit of the United States or any State or any county or any political subdivision of any of the foregoing, whether now or hereafter in existence, having jurisdiction over Tenant or any Property, or any portion thereof, or any Travel Center operated thereon.

- 1.38 "Ground Leases" shall mean, collectively, the ground leases referenced on Exhibit B attached hereto, and any additional ground lease of Property or amendments thereto entered into by Landlord after the Effective Date and accepted by Tenant as being within the scope of the Property and leased to Tenant pursuant to this Agreement.
- 1.39 "Guarantor" shall mean BP Corporation North America Inc., an Indiana corporation, and any successor thereto, replacement thereof or additional guarantor which becomes a Guarantor in accordance with this Agreement.
- 1.40 "Guaranty" shall mean the Second Amended and Restated Guaranty (Third Amended and Restated Lease Agreement No. 2) of even date herewith executed by Guarantor in favor of Landlord, as amended from time to time, and any replacement guaranty or additional guaranty delivered to Landlord with obligations of Tenant first accruing from and after the date of such guaranty (and otherwise in substantially the same form as the Guaranty being delivered by Guarantor to Landlord on the Effective Date) pursuant to this Agreement, as amended from time to time.

1.41 "Hazardous Substances" shall mean any substance:

- (a) the presence of which requires or may hereafter require notification, investigation or remediation under any Applicable Law; or
- (b) which is or becomes defined as a "hazardous waste", "hazardous material" or "hazardous substance" or "pollutant" or "contaminant" under any Applicable Law including, without limitation, the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. Section 9601 et seq.) and the Resource Conservation and Recovery Act (42 U.S.C. Section 6901 et seq.) and the regulations promulgated thereunder; or
- (c) which is toxic, explosive, corrosive, flammable, infectious, radioactive, carcinogenic, mutagenic or otherwise hazardous and is or becomes regulated by any Government Agencies; or
- (d) the presence of which on any Property, or any portion thereof, causes or materially threatens to cause an unlawful nuisance upon such Property, or any portion thereof, or to adjacent properties or poses or materially threatens to pose a hazard to such Property, or any portion thereof, or to the health or safety of persons; or
- (e) without limitation, which contains gasoline, diesel fuel or other petroleum hydrocarbons or volatile organic compounds; or
- (f) without limitation, which contains polychlorinated biphenyls (PCBs) or asbestos or urea formaldehyde foam insulation; or

Salem, OR

- (g) without limitation, which contains or emits radioactive particles, waves or material.
- 1.42 "<u>IFRS</u>" shall mean the international financial reporting standards issued by the International Accounting Standards Board (or successor thereto).
- "Impositions" shall mean, collectively, all taxes (including, without limitation, all taxes imposed under the laws of any State, as such laws may be amended from time to time, and all ad valorem, sales and use, occupancy, or similar taxes as the same relate to or are imposed upon Landlord, Tenant or the business conducted upon the Leased Property), assessments (including, without limitation, all assessments for public improvements or benefit, whether or not commenced or completed prior to the date hereof), water, sewer or other rents and charges, excises, tax levies, fees (including, without limitation, license, permit, inspection, authorization and similar fees), and all other governmental charges, in each case whether general or special, ordinary or extraordinary, foreseen or unforeseen, of every character, in each case in respect of the Leased Property or the business conducted upon the Leased Property by Tenant (including all interest and penalties thereon due to any failure in payment by Tenant), which at any time prior to, during or in respect of the Term hereof may be assessed or imposed on or in respect of or be a lien upon (a) Landlord's interest in the Leased Property, (b) the Leased Property or any part thereof or any rent therefrom or any estate, right, title or interest therein, or (c) any occupancy, operation, use or possession of, or sales from, or activity conducted on, or in connection with the Leased Property or the leasing or use of the Leased Property or any part thereof by Tenant; provided, however, that nothing contained herein shall be construed to require Tenant to pay and the term "Impositions" shall not include (i) any tax based on net income imposed on Landlord, (ii) any net revenue tax of Landlord, (iii) any transfer fee or other tax imposed with respect to the sale, exchange or other disposition by Landlord of the Leased Property or the proceeds thereof, (iv) any single business, gross receipts tax, transaction privilege, rent or similar taxes as the same relate to or are imposed upon Landlord, (v) any interest or penalties imposed on Landlord as a result of the failure of Landlord to file any return or report timely and in the form prescribed by law or to pay any tax or imposition, except to the extent such failure is a result of a breach by Tenant of its obligations pursuant to Section 3.1.2, (vi) any impositions imposed on Landlord that are a result of Landlord not being considered a "United States person" as defined in Section 7701(a)(30) of the Code, (vii) any impositions that are enacted or adopted by their express terms as a substitute for any tax that would not have been payable by Tenant pursuant to the terms of this Agreement, (viii) any impositions imposed as a result of a breach of covenant or representation by Landlord in any agreement governing Landlord's conduct or operation or as a result of the negligence or willful misconduct of Landlord, (ix) any Landlord Lien or any impositions imposed as a result of a transfer or assignment made by Landlord pursuant to Section 21.6.2 or (x) mortgage recording taxes, value added taxes, capital gains taxes or similar taxes, assessments or government levies.
- 1.44 "Insurance Requirements" shall mean all terms of any insurance policy required by this Agreement and all requirements of the issuer of any such policy and all orders, rules and regulations and any other requirements of the National Board of Fire Underwriters (or any other body exercising similar functions) binding upon Landlord, Tenant, any Manager or the Leased Property.

- 1.45 "Interest Rate" shall mean an annual rate of interest, as of the date of Salem, OR determination, equal to the greater of (i) eight and one-half percent (8.5%) per annum and (ii) the per annum rate for ten (10) year U.S. Treasury Obligations as published in The Wall Street Journal plus three hundred fifty (350) basis points.
 - 1.46 "Land" shall have the meaning given such term in Section 2.1(a).
- 1.47 "<u>Landlord</u>" shall have the meaning given such term in the preambles to this Agreement and shall also include their respective permitted successors and assigns.
 - 1.48 "Landlord Default" shall have the meaning given such term in Article 14.
- 1.49 "Landlord Liens" shall mean liens on or against the Leased Property or any payment of Rent (a) which result from any act of, or any claim against, Landlord or any owner of a direct or indirect interest in the Leased Property (other than the lessor under any Ground Lease affecting any portion of the Leased Property), or which result from any violation by Landlord of any terms of this Agreement, or (b) which result from liens in favor of any taxing authority by reason of any tax owed by Landlord or any fee owner of a direct or indirect interest in the Leased Property (other than the lessor under any Ground Lease affecting any portion of the Leased Property); provided, however, that "Landlord Lien" shall not include any lien resulting from any tax for which Tenant is obligated to pay or indemnify Landlord against until such time as Tenant shall have already paid to or on behalf of Landlord the tax or the required indemnity with respect to the same.
- 1.50 "<u>Lease Year</u>" shall mean each consecutive full twelve calendar month period during the Term, except that if the Effective Date shall not occur on the first day of a month, the first Lease Year shall be for the period from the Effective Date through the last day of the month in which the first anniversary of the Effective Date shall occur.
- 1.51 "<u>Leased Improvements</u>" shall have the meaning given such term in <u>Section</u> 2.1(b).
 - 1.52 "Leased Property" shall have the meaning given such term in Section 2.1.
- 1.53 "Legal Requirements" shall mean all federal, state, county, municipal and other governmental statutes, laws, rules, orders, regulations, ordinances, judgments, decrees and injunctions affecting the Leased Property or the maintenance, construction, alteration or operation thereof, whether now or hereafter enacted or in existence, including, without limitation, (a) all permits, licenses, authorizations and regulations necessary to operate any Property for its Permitted Use, and (b) all covenants, agreements, restrictions and encumbrances contained in any instruments at any time in force affecting any Property, including those which may (i) require material repairs, modifications or alterations in or to any Property or (ii) in any way materially and adversely affect the use and enjoyment thereof, but excluding any requirements arising as a result of Landlord's status as a real estate investment trust.
- 1.54 "<u>Lien</u>" shall mean any mortgage, security interest, pledge, collateral assignment, or other encumbrance, lien or charge of any kind, or any transfer of property or assets for the

purpose of subjecting the same to the payment of indebtedness or performance of any other obligation in priority to payment of general creditors.

- 1.55 "Manager" shall mean, with respect to any Property, any operator or manager engaged by Tenant from time to time to operate or manage such Property.
- 1.56 "Merger" shall have the meaning given such term in the recitals to this Agreement.
- 1.57 "Minerals" shall mean all the oil, gas, associated hydrocarbons, lead, zinc, copper, coal, coal seam gas, coalbed methane, lignite, peat, sulphur, phosphate, iron ore, sodium, salt, uranium, thorium and other fissionable materials, molybdenum, vanadium, titanium, ilmenite, rutile, leucoxene, zircon, gold, silver, platinum, palladium, bauxite, granite, limestone, bedrock of any kind or character, kaolin and other clays, sand, gravel, construction aggregate and other mined or quarried stone or rock material, industrial minerals, geothermal energy, and all other substances and ore deposits of any kind or character, whether solid, liquid or gaseous, and without limitation by enumeration of the minerals and substances expressly mentioned above, in, on or under any part of the Real Property.
- 1.58 "Minimum Net Worth" shall mean Fifteen Billion U.S. Dollars (\$15,000,000,000).
- 1.59 "Minimum Rent" shall mean amounts due as provided in the Prior Lease, and, as of the Effective Date, shall mean, for the first Lease Year, Forty-Six Million One Hundred Ten Thousand Eight Hundred Seventeen and 22/100ths Dollars (\$46,110,817.22) per annum, and, for each subsequent Lease Year, Minimum Rent shall be the product of the Minimum Rent for the immediately preceding Lease Year multiplied by 1.02.
- 1.60 "Minimum Rent Reduction Limit" shall mean, for the first Lease Year, \$2,305,541, and, for each subsequent Lease Year, the Minimum Rent Reduction Limit shall be an amount equal to the product of (a) the Minimum Rent Reduction Limit for the immediately preceding Lease Year, multiplied by (b) 1.02.
 - 1.61 "Net Worth" shall mean Total Assets less Total Liabilities.
- 1.62 "Non-Recourse Parties" shall have the meaning given such term in Section 24.17.
 - 1.63 "Notice" shall mean a notice given in accordance with Section 24.10.
 - 1.64 "Offer" shall have the meaning given such term in Section 4.1.1(b).
 - 1.65 "Offer Notice" shall have the meaning given such term in Section 2.6.
 - 1.66 "Offered Property" shall have the meaning given such term in Section 2.6.

Salem, OR

1.67 "Officer's Certificate" shall mean a certificate signed by an officer or other duly authorized individual of the certifying Entity duly authorized by the board of directors or other governing body of the certifying Entity.

1, 1511

- 1.68 "Operating Rights" shall have the meaning given such term in Section 5.3.
- 1.69 "Other Leases" shall mean, collectively, (a) that certain Third Amended and Restated Lease Agreement No. 1, dated as of the Effective Date, between Landlord and Tenant, together with all modifications, amendments and supplements thereto, (b) that certain Third Amended and Restated Lease Agreement No. 3, dated as of the Effective Date, between Landlord and Tenant, together with all modifications, amendments and supplements thereto, (c) that certain Third Amended and Restated Lease Agreement No. 4, dated as of the Effective Date, between Landlord and Tenant, together with all modifications, amendments and supplements thereto, and (d) that certain Second Amended and Restated Lease Agreement No. 5, dated as of the Effective Date, between Highway Ventures Properties Trust, Highway Ventures Properties LLC and Tenant, together with all modifications, amendments and supplements thereto.
- 1.70 "Overdue Rate" shall mean, on any date, a per annum rate of interest equal to the lesser of the Interest Rate plus four percent (4%) and the maximum rate then permitted under applicable law.
- 1.71 "<u>Parent</u>" shall have the meaning given to such term in the recitals to this Agreement.
 - 1.72 "Percentage Reduction" shall be eight and one-half percent (8.5%).
- 1.73 "Permitted Encumbrances" shall mean, with respect to any Property, all rights, restrictions, and easements of record set forth on Schedule B to the applicable owner's or leasehold title insurance policy issued to Landlord with respect to such Property, plus any other encumbrances as may have been granted or caused by Landlord or otherwise consented to in writing by Landlord from time to time.
- 1.74 "Permitted Use" shall mean, with respect to any Property, any use of such Property permitted pursuant to Section 4.1.1.
- 1.75 "Person" shall mean any individual or Entity, and the heirs, executors, administrators, legal representatives, successors and assigns of such Person where the context so admits.
- 1.76 "<u>Prior Lease</u>" shall have the meaning given such term in the recitals to this Agreement.
 - 1.77 "Property" shall have the meaning given such term in Section 2.1.
- 1.78 "<u>Property Mortgage</u>" shall mean any Encumbrance placed upon the Leased Property, or any portion thereof, in accordance with <u>Article 20</u>.
 - 1.79 "Property Mortgagee" shall mean the holder of any Property Mortgage.

- 1.80 "Qualifying Guarantor" shall mean a Person that satisfies one or more of the following:
 - (a) such Person is rated at least "BBB-" by S&P Global Ratings, acting through Standard and Poor's Financial Services LLC, or "Baa3" by Moody's Investors Services, Inc.; or
 - (b) such Person has a Net Worth equal to or greater than the Minimum Net Worth; or
 - (c) such Person has a creditworthiness that is acceptable to Landlord in its sole discretion.

Any Qualifying Guarantor under clause (b) shall provide to Landlord, (i) prior to becoming Guarantor, such Person's audited financial statement prepared in accordance with GAAP and certified by an independent certified public accountant, for such Person's most recently completed fiscal year (or, if the date of determination is on or before the date that falls within ninety (90) days after the end of such most recently completed fiscal year and the audit of such fiscal year has not yet been released, then, the audited financial statement of such Person at the end of the fiscal year immediately preceding such most recently completed fiscal year) and, (ii) if such Qualifying Guarantor is not a BP Affiliate, for each subsequent fiscal year, such Person's audited financial statement for such Person's most recent fiscal year, prepared in accordance with GAAP and certified by an independent certified public accountant, within thirty (30) days of the release of such annual statement for such fiscal year but not more than eighteen (18) months after delivery of the prior statement, demonstrating that such Person had the Minimum Net Worth at the end of that fiscal year.

- 1.81 "Real Property" shall have the meaning given such term in Section 2.1.
- 1.82 "Related Person" shall have the meaning given such term in Section 24.15.
- 1.83 "Rent" shall mean, collectively, the Minimum Rent and Additional Charges.
- 1.84 "Rules" shall have the meaning given such term in Section 22.1.
- 1.85 "<u>SARA</u>" shall mean the Superfund Amendments and Reauthorization Act of 1986, as the same has been and may be amended, restated, modified or supplemented from time to time.
 - 1.86 "SEC" shall mean the Securities and Exchange Commission.
- 1.87 "Shell" shall mean Equilon Enterprises LLC (doing business as Shell Oil Products US), a Delaware limited liability company.
- 1.88 "Shell Agreement" shall mean that certain Liquefied Natural Gas Dispensing Site License and Sales Agreement, dated as of April 15, 2013, between Tenant and Shell, together with all modifications, amendments and supplements thereto.

- 1.89 "Shell SNDA" shall have the meaning given such term in Section 4.5. Salem, OR
- 1.90 "State" shall mean, with respect to any Property, the state, commonwealth or district in which such Property is located.
 - 1.91 "Successor Landlord" shall have the meaning given such term in Section 20.2.
 - 1.92 "Superior Landlord" shall have the meaning given such term in Section 20.2.
 - 1.93 "Superior Lease" shall have the meaning given such term in Section 20.2.
 - 1.94 "Superior Mortgage" shall have the meaning given such term in Section 20.2.
 - 1.95 "Superior Mortgagee" shall have the meaning given such term in Section 20.2.
- 1.96 "<u>SVC</u>" shall mean Service Properties Trust, a Maryland real estate investment trust.
 - 1.97 "TCA" shall have the meaning given such term in the recitals hereto.
- 1.98 "<u>Tenant</u>" shall have the meaning given such term in the preambles to this Agreement and shall also include its permitted successors and assigns.
- 1.99 "Tenant's Personal Property" shall mean all motor vehicles and consumable inventory and supplies, furniture, furnishings, equipment, movable walls and partitions, equipment and machinery and all other tangible personal property of Tenant acquired by Tenant before, on or after the Effective Date and located at the Leased Property or used in Tenant's business at the Leased Property and all modifications, replacements, alterations and additions to such personal property installed at the expense of Tenant, other than any items included within the definition of Fixtures, and any items which are not fixtures and which Tenant shall have (x) purchased after the Effective Date and (y) removed from the Leased Property by the expiration or earlier termination of the Term.
- 1.100 "<u>Term</u>" shall mean the period commencing on the Effective Date and ending at the later of (a) the expiry of the Fixed Term and (b) if Tenant has properly exercised its right to extend the Term as provided in Section 2.4, the expiry of the then applicable Extended Term, in either case unless sooner terminated pursuant to the provisions of this Agreement.
- 1.101 "<u>Total Assets</u>" shall mean, as at any date of determination, all assets of the Guarantor and its subsidiaries determined on a consolidated basis in conformity with GAAP.
- 1.102 "<u>Total Liabilities</u>" shall mean, as at any date of determination, all liabilities of the Guarantor and its subsidiaries on a consolidated basis in conformity with GAAP.
- 1.103 "Travel Center" shall mean, with respect to any Property, collectively, the hospitality, fuel and service facilities located at such Property, including hotel, food and beverage services facilities, fuel pumps, facilities for the storage and distribution of petroleum products and storage, generation and distribution of other fuels or energy sources (including,

without limitation, electricity), retail shops and other facilities or services being operated or from time to time proposed to be operated on such Property.

1.104 "Unsuitable for Its Permitted Use" shall mean, with respect to any Travel Center, a state or condition such that following any damage, destruction or Condemnation, such Travel Center cannot be operated on a commercially practicable basis for its Permitted Use and it cannot reasonably be expected to be restored to substantially the same condition as existed immediately before such damage, destruction or Condemnation, and as otherwise required by this Agreement, within twenty-four (24) months following such damage, destruction or Condemnation or such longer period of time as to which business interruption insurance or Award proceeds is available to cover Rent and other costs related to the applicable Property following such damage, destruction or Condemnation.

ARTICLE 2 LEASED PROPERTY AND TERM

- 2.1 <u>Leased Property</u>. Upon and subject to the terms and conditions hereinafter set forth, Landlord leases to Tenant and Tenant leases from Landlord all of Landlord's right, title and interest in and to all of the following (each of items (a) through (e) below which, as of the Effective Date, relates to any single Travel Center, a "<u>Property</u>", and collectively, the "<u>Leased Property</u>", and those portions of the Leased Property described in items (a) through (d) below being the "<u>Real Property</u>"):
 - (a) those certain tracts, pieces and parcels of land, as more particularly described in Exhibits A-1 through A-36 attached hereto and made a part hereof (the "Land");
 - (b) all buildings, structures and other improvements of every kind including, but not limited to, underground storage tanks, alleyways and connecting tunnels, sidewalks, utility pipes, conduits and lines (on-site and off-site), parking areas and roadways appurtenant to such buildings and structures presently situated upon the Land (collectively, the "Leased Improvements");
 - (c) all easements, rights and appurtenances relating to the Land and the Leased Improvements, but excluding all of Landlord's right, title and interest in and to all Minerals and all executory rights and other rights necessary to sell, lease or otherwise convey the Minerals, all of which are expressly reserved by Landlord, provided, however, that Landlord shall not, and shall not authorize or permit any others to, conduct any exploration, evaluation or extraction of any Minerals or pursue any other similar activities relating to the Minerals during the Term;
 - (d) all equipment, machinery and fixtures integral to the operation of the Leased Improvements, and other items of property now or hereafter permanently affixed or integral to or incorporated into the Leased Improvements, including, without limitation, all furnaces, boilers, heaters, electrical equipment, heating, plumbing, lighting, ventilating, refrigerating, incineration, air and water pollution control, waste disposal, aircooling and air-conditioning systems and apparatus, sprinkler systems and fire and theft

SEP 0 2 2025

protection equipment, all of which, to the maximum extent permitted by law, are hereby deemed by the parties hereto to constitute real estate, together with all replacements, modifications, alterations and additions thereto, but specifically excluding all items included within the category of Tenant's Personal Property (collectively, the "Fixtures"); and

- (e) any and all leases of space in the Leased Improvements, including, without limitation, the Ground Leases.
- 2.2 Condition of Leased Property. Tenant acknowledges that it is and has been in possession of the Leased Property, and Tenant accepted the Leased Property in its "as is" condition, subject to the rights of parties in possession, the existing state of title, including all covenants, conditions, restrictions, reservations, mineral leases, easements and other matters of record or that are visible or apparent on the Leased Property, all applicable Legal Requirements, the lien of any financing instruments, mortgages and deeds of trust existing prior to the Effective Date or permitted by the terms of this Agreement, and such other matters which would be disclosed by an inspection of the Leased Property and the record title thereto or by an accurate survey thereof. TENANT REPRESENTS THAT IT IS NOT RELYING ON ANY REPRESENTATION OR WARRANTY OF LANDLORD OR LANDLORD'S AGENTS OR EMPLOYEES WITH RESPECT THERETO AND TENANT WAIVES ANY CLAIM OR ACTION AGAINST LANDLORD IN RESPECT OF THE CONDITION OF THE LEASED PROPERTY. LANDLORD MAKES NO WARRANTY OR REPRESENTATION, EXPRESS OR IMPLIED, IN RESPECT OF THE LEASED PROPERTY OR ANY PART THEREOF. EITHER AS TO ITS FITNESS FOR USE, DESIGN OR CONDITION FOR ANY PARTICULAR USE OR PURPOSE OR OTHERWISE, AS TO THE QUALITY OF THE MATERIAL OR WORKMANSHIP THEREIN, LATENT OR PATENT, IT BEING AGREED THAT ALL SUCH RISKS ARE TO BE BORNE BY TENANT. To the maximum extent permitted by law, however, Landlord hereby assigns to Tenant all of Landlord's rights to proceed against any predecessor in interest or insurer for breaches of warranties or representations or for latent defects in the Leased Property. Landlord shall fully cooperate with Tenant in the prosecution of any such claims, in Landlord's or Tenant's name, all at Tenant's sole cost and expense. Tenant shall indemnify, defend, and hold harmless Landlord from and against any loss, cost, damage or liability (including reasonable attorneys' fees) incurred by Landlord in connection with such cooperation.
- 2.3 <u>Term.</u> The initial term of this Agreement (the "<u>Fixed Term</u>") shall expire on the day preceding the tenth anniversary of the Effective Date.
- 2.4 Extended Terms. Tenant shall have the right to extend the Term for up to five (5) consecutive renewal terms of ten (10) years each (each, an "Extended Term"), provided that (unless Landlord shall elect in its sole discretion to waive any such condition), at the time Tenant exercises a right to extend the Term, (a) no Event of Default shall have occurred and be continuing and (b) on the date of the exercise of the extension option and on the first day of each such Extended Term, there shall be a Guaranty in favor of Landlord from a Qualifying Guarantor which shall either meet the standard in clause (a) or (c) of the definition of Qualifying Guarantor or shall provide Landlord with the audited financial statement described in the last paragraph of

the definition of Qualifying Guarantor to establish that it meets the criteria in clause (b) thereof to be a Qualifying Guarantor, even if such Entity is a BP Affiliate.

All of the terms, covenants and provisions of this Agreement shall apply to each Extended Term (including, but without limitation, the two percent annual Minimum Rent increases provided in the definition of Minimum Rent), except that Tenant shall have no right to extend the Term beyond the expiration of the fifth Extended Term. If Tenant shall elect to exercise its option to extend the Term for any Extended Term, it shall do so by giving Landlord Notice thereof not later than eighteen (18) months prior to the commencement of the applicable Extended Term, it being understood and agreed that time shall be of the essence with respect to the giving of any such Notice. If Tenant shall fail to give any such Notice, this Agreement shall automatically terminate at the end of the Fixed Term or the applicable Extended Term, and Tenant shall have no further option to extend the Term of this Agreement. If Tenant shall give such Notice, the extension of this Agreement shall be automatically effected without the execution of any additional documents; it being understood and agreed, however, that Tenant and Landlord shall execute such documents and agreements as either party shall reasonably require to evidence the same.

- Right to Repool Properties. Landlord shall have the right from time to time in connection with a financing or other capital raising transaction to terminate the Term of this Agreement with respect to one or more Properties and contemporaneously to lease such Properties back to Tenant or an Affiliated Person as to Tenant under one of the Other Leases, and/or one or more new lease(s) as determined by Landlord (a "repooling"), provided that Landlord shall have obtained Tenant's prior written consent to any such repooling, which consent shall not be unreasonably withheld, conditioned or delayed; provided, however, that Tenant may grant or withhold consent in Tenant's sole discretion if a proposed repooling, after taking into account all prior repoolings, if any, would result in the repooling of a number of Properties that would exceed fifty percent (50%) of the Properties hereunder as of the Effective Date (i.e., a repooling of more than eighteen (18) Travel Centers in total over the Term) or would result in one or more new leases (as opposed to adding a Property to one of the Other Leases). Each party agrees to execute and deliver such documentation as the other party may reasonably request in connection with any such new lease or repooling, including, without limitation, a new lease, a lease amendment, and a new guaranty from Guarantor or confirmation from Guarantor that its existing Guaranty applies to any such new lease or lease amendment.
- 2.6 Right of First Offer. So long as this Agreement is still in full force and effect and there then exists no Event of Default, if Landlord intends to solicit offers, or to accept an unsolicited offer, to purchase its fee interest or leasehold interest in any Property, Landlord first shall offer to sell the applicable Property or its leasehold interest under a Ground Lease (the "Offered Property") to Tenant at a price to be identified by Landlord in such offer notice (the "Offer Notice"), which Offer Notice shall also include the material terms on which Landlord is offering to sell the Offered Property to Tenant and the estimated closing date. If Tenant shall elect to accept such offer, it shall give Landlord notice of such election within twenty (20) Business Days after the Offer Notice is given. If Tenant elects to accept such offer within such twenty (20) Business Day period, Landlord and Tenant shall, for a period of thirty (30) days after the date of Tenant's election, engage in good faith negotiations of a mutually acceptable purchase and sale agreement incorporating the terms and conditions in Landlord's Offer Notice

and such other terms as are necessary for the transaction and agreed by the parties, acting reasonably. In the event Tenant fails to accept Landlord's offer within such twenty (20) Business Day period, or the parties, having negotiated in good faith, fail to execute and deliver a mutually acceptable purchase and sale agreement within such thirty (30) day period, then Landlord shall have the right to accept an offer and/or enter into an agreement to sell and/or to sell such Offered Property to a third party, provided, however, that: (a) Landlord's conveyance of the Offered Property shall take place within two hundred seventy (270) days of delivery of the Offer Notice; (b) the purchase price paid for the Offered Property shall be equal to or greater than 97% of the purchase price included in the Offer Notice; and (c) the other terms of such sale taken as a whole shall be substantially the same or better for Landlord than the proposed terms contained in the Offer Notice. Tenant's rights shall be reinstituted with respect to such Offered Property if Landlord shall not so convey title to the Offered Property to a third party within two hundred seventy (270) days following delivery of the Offer Notice. Tenant's rights under this paragraph shall not apply to (i) the grant of a mortgage lien as collateral in connection with a bona fide financing, (ii) a foreclosure sale or deed in lieu thereof with respect to bona fide thirdparty indebtedness (but not, for the avoidance of doubt, any subsequent transfers of such Offered Property by such foreclosing lender or its designee), (iii) a transfer to any entity that is a Controlled subsidiary or Controlling parent of, or an entity under common Control with, Landlord, (iv) a transfer to any entity whose business is managed by The RMR Group LLC, The RMR Group Inc., or any Controlled subsidiary or Controlling Entity of any of the foregoing, or (v) a sale or other transfer pursuant to or in lieu of taking by eminent domain.

ARTICLE 3 RENT

- Rent. Tenant shall pay, in lawful money of the United States of America which shall be legal tender for the payment of public and private debts, without offset, abatement, demand or deduction (unless otherwise expressly provided in this Agreement), Minimum Rent to Landlord and Additional Charges to the party to whom such Additional Charges are payable, during the Term. All payments to Landlord shall be made by wire transfer of immediately available federal funds or by other means acceptable to Landlord in its sole discretion. Rent hereunder and under the Prior Lease for any partial month shall be prorated on a per diem basis.
- 3.1.1 <u>Minimum Rent</u>. Minimum Rent shall be paid in equal monthly installments in advance on the first Business Day of each calendar month during the Term.
- 3.1.2 <u>Additional Charges</u>. In addition to the Minimum Rent payable hereunder, Tenant shall pay (or cause to be paid) to the appropriate parties and discharge (or cause to be discharged) as and when due and payable the following (collectively, "<u>Additional Charges</u>"):
 - (a) <u>Impositions</u>. Subject to <u>Article 8</u> relating to permitted contests, Tenant shall pay, or cause to be paid, all Impositions before any fine, penalty, interest or cost (other than any opportunity cost as a result of a failure to take advantage of any discount for early payment) may be added for non-payment, such payments to be made directly to the taxing authorities where feasible, and shall promptly, upon request, furnish to Landlord copies of official receipts or other reasonably satisfactory proof evidencing

such payments. If any such Imposition may, at the option of the taxpayer, lawfully be paid in installments (whether or not interest shall accrue on the unpaid balance of such Imposition), Tenant may exercise the option to pay the same (and any accrued interest on the unpaid balance of such Imposition) in installments and, in such event, shall pay, or cause to pay, such installments during the Term as the same become due and before any fine, penalty, premium, further interest or cost may be added thereto. Landlord, at its expense, shall, to the extent required or permitted by Applicable Law, prepare and file, or cause to be prepared and filed, all tax returns and pay all taxes due in respect of Landlord's net income, gross receipts, sales and use, single business, transaction privilege, rent, ad valorem, franchise taxes and taxes on its capital stock or other equity interests, and Tenant, at its expense, shall, to the extent required or permitted by Applicable Laws and regulations, prepare and file all other tax returns and reports in respect of any Imposition as may be required by Government Agencies. If any refund shall be due from any taxing authority in respect of any Imposition paid by or on behalf of Tenant, the same shall be paid over to or retained by Tenant. Landlord and Tenant shall, upon request of the other, provide such data as is maintained by the party to whom the request is made with respect to the Leased Property as may be necessary to prepare any required returns and reports. In the event Government Agencies classify any property covered by this Agreement as personal property, Tenant shall file, or cause to be filed, all personal property tax returns in such jurisdictions where it may legally so file. Each party shall, to the extent it possesses the same, provide the other, upon request, with cost and depreciation records necessary for filing returns for any property so classified as personal property. Where Landlord is legally required to file personal property tax returns for property covered by this Agreement, Landlord shall provide Tenant with copies of assessment notices in sufficient time for Tenant to file a protest. All Impositions assessed against such personal property shall be (irrespective of whether Landlord or Tenant shall file the relevant return) paid by Tenant not later than the last date on which the same may be made without interest or penalty, subject to the provisions of Article 8. Landlord shall give prompt Notice to Tenant of all Impositions payable by Tenant hereunder of which Landlord at any time has knowledge; provided, however, that Landlord's failure to give any such notice shall in no way diminish Tenant's obligation hereunder to pay such Impositions.

- (b) <u>Utility Charges</u>. All charges for electricity, power, gas, oil, water and other utilities used in connection with the Leased Property.
- (c) <u>Insurance Premiums</u>. All premiums for the insurance coverage required to be maintained pursuant to <u>Article 9</u>.
- (d) Other Charges. All other amounts, liabilities and obligations payable by Tenant under this Agreement.
- 3.1.3 <u>Reimbursement for Additional Charges</u>. If Tenant pays or causes to be paid property taxes or similar or other Additional Charges attributable to periods after the end of the Term, whether upon expiration or sooner termination of this Agreement, Tenant may, within a reasonable time after the end of the Term, provide Notice to Landlord of its estimate of such

amounts. Landlord shall promptly reimburse Tenant for all payments of such taxes and other similar Additional Charges that are attributable to any period after the Term of this Agreement.

- Late Payment of Rent, Etc. If any installment of Minimum Rent shall not be paid when due hereunder or any Additional Charges (but only as to those Additional Charges which are payable directly to Landlord) shall not be paid within ten (10) Business Days of the date the same are due hereunder, Tenant shall pay Landlord, on demand, as Additional Charges, a late charge (to the extent permitted by law) computed at the Overdue Rate on the amount of such installment, from the due date of such installment or amount to the date of payment thereof. To the extent that Tenant pays any Additional Charges directly to Landlord or any Property Mortgagee or lessor, as applicable, pursuant to any requirement of this Agreement, Tenant shall be relieved of its obligation to pay such Additional Charges to the Entity to which they would otherwise be due. If any payments due from Landlord to Tenant shall not be paid within ten (10) days after its due date, Landlord shall pay to Tenant, on demand, a late charge (to the extent permitted by law) computed at the Overdue Rate on the amount of such installment from the due date of such installment to the date of payment thereof. In the event of any failure by Tenant to pay any Additional Charges when due, Tenant shall promptly pay and discharge, as Additional Charges, every fine, penalty, interest and cost which is added for non-payment or late payment of such items. Landlord shall have all legal, equitable and contractual rights, powers and remedies provided either in this Agreement or by statute or otherwise in the case of non-payment of the Additional Charges as in the case of non-payment of the Minimum Rent.
- 3.3 Net Lease, Etc. The Rent shall be absolutely net to Landlord so that this Agreement shall yield to Landlord the full amount of the installments or amounts of the Rent throughout the Term, subject to any other provisions of this Agreement which expressly provide otherwise, including those provisions for adjustment or abatement of such Rent. Landlord and Tenant acknowledge and agree that none of the Rent provided for under this Agreement is allocable to any personal property included in the Leased Property.
- 3.4 No Termination, Abatement, Etc. Except as otherwise specifically provided in this Agreement, each of Landlord and Tenant, to the maximum extent permitted by law, shall remain bound by this Agreement in accordance with its terms and shall not take any action without the consent of the other to modify, surrender or terminate this Agreement. In addition, except as otherwise expressly provided in this Agreement, Tenant shall not seek, or be entitled to, any abatement, deduction, deferment or reduction of the Rent, or set-off against the Rent, nor shall the respective obligations of Landlord and Tenant be otherwise affected by reason of (a) any damage to or destruction of the Leased Property, or any portion thereof, from whatever cause or any Condemnation; (b) the lawful or unlawful prohibition of, or restriction upon, Tenant's use of the Leased Property, or any portion thereof, or the interference with such use by any Person other than Landlord or any Person acting through Landlord, (c) eviction by fee owners of any Properties due to termination of any Ground Leases; (d) any claim which Tenant may have against Landlord by reason of any default (other than a monetary default) or breach of any warranty by Landlord under this Agreement or any other agreement between Landlord and Tenant, or to which Landlord and Tenant are parties (but, for the avoidance of doubt, excluding any determination that Landlord does not have fee title to any Property not subject to a Ground Lease); (e) any bankruptcy, insolvency, reorganization, composition, readjustment, liquidation, dissolution, winding up or other proceedings affecting Landlord or any assignee or transferee of

Landlord; or (f) for any other cause whether similar or dissimilar to any of the foregoing (other than a monetary default by Landlord). Except as otherwise specifically provided in this Agreement, Tenant hereby waives all rights arising from any occurrence whatsoever, which may now or hereafter be conferred upon it by law (1) to modify, surrender or terminate this Agreement or quit or surrender the Leased Property, or any portion thereof, or (2) which would entitle Tenant to any abatement, reduction, suspension or deferment of the Rent or other sums payable or other obligations to be performed by Tenant hereunder. The obligations of Tenant hereunder shall be separate and independent covenants and agreements, and the Rent and all other sums payable by Tenant hereunder shall continue to be payable in all events unless this Agreement is terminated or the obligations to pay the same shall be terminated pursuant to the express provisions of this Agreement.

3.5 **Prepayment.** Landlord acknowledges that, concurrent with the Effective Date, Tenant has prepaid Minimum Rent in the amount of \$34,129,266.29. Accordingly, Landlord agrees that Tenant shall be entitled to a credit of \$378,205.52 against each monthly payment of Minimum Rent due under this Agreement on or after the Effective Date through the end of the Fixed Term. If this Agreement terminates prior to its scheduled expiration, any such unapplied credit shall be retained by the Landlord as Additional Rent.

ARTICLE 4 USE OF THE LEASED PROPERTY

4.1 **Permitted Use.**

4.1.1 Permitted Use.

Tenant shall, at all times during the Term, and at any other time that Tenant shall be in possession of any Property, continuously use and operate, or cause to be used and operated, such Property as a Travel Center and any uses incidental thereto, and any truck servicing or repair, retail convenience, mobility or energy generation or management facility and any other ancillary lawful uses related or complimentary thereto; provided, however, Tenant may from time to time suspend use or operations at any Property as in Tenant's reasonable determination is necessary or desirable in connection with construction or development thereat, casualty or condemnation with respect thereto, in order to comply with Applicable Law or, if, in Tenant's reasonable determination, it is no longer economically practical to operate such Property as currently operated. Tenant shall not use (and shall not permit any Person to use) any Property, or any portion thereof, for any other use without the prior written consent of Landlord, which approval shall not be unreasonably withheld, delayed or conditioned. No use shall be made or permitted to be made of any Property and no acts shall be done thereon which will cause the cancellation of any insurance policy covering such Property or any part thereof (unless another adequate policy is available) or which would constitute a default under any Ground Lease affecting such Property, nor shall Tenant sell or otherwise provide, or permit to be kept, used or sold in or about any Property any article which may be prohibited by law or by the standard form of fire insurance policies, or any other insurance policies required to be carried hereunder, or fire underwriter's regulations. Tenant shall, at its sole cost, comply or cause to be complied with all Insurance

Requirements. Tenant shall not take or omit to take, or permit to be taken or omitted to be taken, any action, the taking or omission of which materially impairs the value or the usefulness of any Property or any part thereof for its Permitted Use.

- (b) In the event that, in the reasonable determination of Tenant, it shall no longer be economically practical to operate any Property as a Travel Center, Tenant shall give Landlord Notice thereof, which Notice shall set forth in reasonable detail the reasons therefor. Thereafter, Landlord and Tenant shall negotiate in good faith to agree on an alternative use for such Property; provided, however, in no event shall the Minimum Rent be reduced or abated as a result thereof. If Landlord and Tenant fail to agree on an alternative use for such Property within sixty (60) days after commencing negotiations as aforesaid, Tenant may market such Property for sale to a third party. If Tenant receives a bona fide offer (an "Offer") to purchase such Property from a Person having the financial capacity to implement the terms of such Offer. Tenant shall give Landlord Notice thereof, which Notice shall include a copy of the Offer executed by such third party. In the event that Landlord shall fail to accept or reject such Offer within thirty (30) days after receipt of such Notice, such Offer shall be deemed to be rejected by Landlord. If Landlord shall sell the Property pursuant to such Offer, then, effective as of the date of such sale, this Agreement shall terminate with respect to such Property, and the Minimum Rent shall be reduced by an annual amount equal to the product of (x) the Percentage Reduction multiplied by (y) the net proceeds of sale received by Landlord. If Landlord shall reject (or be deemed to have rejected) such Offer, then, effective as of the proposed date of such sale, this Agreement shall terminate with respect to such Property, and the Minimum Rent shall be reduced by an annual amount equal to the product of (a) the Percentage Reduction multiplied by (b) the projected net proceeds determined by reference to such Offer; provided, however, in no event shall the aggregate reduction of Minimum Rent during the Term pursuant to this Section 4.1.1(b) exceed the Minimum Rent Reduction Limit.
- 4.1.2 <u>Necessary Approvals</u>. Tenant shall proceed with all due diligence and exercise reasonable efforts to obtain and maintain, or cause to be obtained and maintained, all approvals necessary to use and operate, for its Permitted Use, each Property and the Travel Center located thereon under applicable law.
- 4.1.3 <u>Lawful Use, Etc.</u> Tenant shall not, and shall not permit any Person to, use or suffer or permit the use of any Property or Tenant's Personal Property, if any, for any unlawful purpose. Tenant shall not, and shall not permit any Person to, commit or suffer to be committed any waste on any Property, or in any Travel Center, nor shall Tenant cause or permit any unlawful nuisance thereon or therein. Tenant shall not, and shall not permit any Person to, suffer nor permit any Property, or any portion thereof, to be used in such a manner as (i) may materially and adversely impair Landlord's or Tenant's title thereto or to any portion thereof, or (ii) may reasonably allow a claim or claims for adverse usage or adverse possession by the public, as such, or of implied dedication of such Property, or any portion thereof.
- 4.1.4 <u>Compliance with Legal/Insurance Requirements, Etc.</u> Subject to the provisions of <u>Article 8</u>, Tenant, at its sole expense, shall (i) comply with (or cause to be complied with) all material Legal Requirements and Insurance Requirements in respect of the use,

operation, maintenance, repair, alteration and restoration of any Property and with the terms and conditions of any Ground Lease affecting any Property, (ii) perform (or cause to be performed) in a timely fashion all of Landlord's obligations under any Ground Lease affecting any Property and (iii) procure, maintain and comply with (or cause to be procured, maintained and complied with) all material licenses, permits and other authorizations and agreements required for any use of any Property and Tenant's Personal Property, if any, then being made, and for the proper erection, installation, operation and maintenance of the Leased Property or any part thereof.

4.2 Environmental Matters.

4.2.1 Restriction on Use, Etc. During the Term and any other time that Tenant shall be in possession of any Property, Tenant shall not, and shall not permit any Person to, store on, release or spill upon, dispose of or transfer to or from such Property any Hazardous Substance, except in compliance with all Applicable Laws in all material respects. During the Term and any other time that Tenant shall be in possession of any Property, Tenant shall maintain (or shall cause to be maintained) such Property at all times free of any Hazardous Substance (except in compliance with all Applicable Laws). Tenant shall promptly (and shall direct any Manager to promptly): (a) upon receipt of notice or knowledge, notify Landlord in writing of any material change in the nature or extent of Hazardous Substances at any Property, (b) transmit to Landlord a copy of any report which is required to be filed by Tenant or any Manager with respect to any Property pursuant to SARA Title III or any other similar Applicable Law, (c) transmit to Landlord copies of any citations, orders, notices or other governmental communications received by Tenant or any Manager or their respective agents or representatives with respect to Hazardous Substances violations or alleged violations of Applicable Law (each an "Environmental Notice"), which Environmental Notice requires a written response or any action to be taken and/or if such Environmental Notice gives notice of and/or presents a material risk of any material violation of any Applicable Law and/or presents a material risk of any material cost, expense, loss or damage (an "Environmental Obligation"), (d) observe and comply with (or cause to be observed and complied with) all material Applicable Laws relating to the use, storage, maintenance and disposal of Hazardous Substances and all orders or directives from any official, court or agency of competent jurisdiction relating to the use, storage or maintenance, or requiring the removal, treatment, containment or other disposition of Hazardous Substances, and (e) pay or otherwise dispose (or cause to be paid or otherwise disposed) of any fine, charge or Imposition related to Hazardous Substances or violations of Applicable Law for which Tenant or any Person claiming by, through or under Tenant and/or Landlord are legally liable, unless Tenant or any Manager shall contest the same in good faith and by appropriate proceedings and the right to use and the value of any of the Leased Property is not materially and adversely affected thereby.

If, at any time prior to the termination of this Agreement, Hazardous Substances (other than those maintained in accordance with Applicable Laws) are discovered on any Property, subject to Tenant's right to contest the same in accordance with Article 8, Tenant shall take (and shall cause to be taken) all actions and incur any and all expenses, as are required by any Government Agency and by Applicable Law, (i) to clean up and remove from and about such Property all Hazardous Substances thereon, (ii) to contain and prevent any further discharge, release or threat of discharge or release of Hazardous Substances on or about such Property and

Salem. OR

- (iii) to use good faith efforts to eliminate any further discharge, release or threat of discharge or release of Hazardous Substances on or about such Property.
- 4.2.2 Environmental Report. Tenant shall, at its sole cost and expense, provide Landlord, not more than six (6) months before the scheduled expiration (or, if applicable, within twelve (12) months following any earlier termination) of the Term, with an Environmental Report (as hereinafter defined), prepared by an environmental consultant reasonably acceptable to Landlord and dated within six (6) months of the scheduled expiration (or in the case of an earlier termination of this Agreement, within twelve (12) months after such termination) concluding, if true, and subject to customary limitations and standards, that the Leased Property does not (or, if delivered after the Term, at the end of the Term did not) contain any "recognized environmental condition" (as defined in the most recent version of the ASTM standard practice for Phase I environmental site assessments) other than in compliance with Applicable Law, or such other generally accepted standard then in use for commercial transactions. An "Environmental Report" shall be a so-called "Phase I" report or such other level of investigation which shall be the standard of diligence in the purchase or lease of similar property at the time, together with any additional investigation and report which would be needed to make the conclusions required above or which would customarily follow any discovery contained in any initial report(s), and for which the investigation and testing on which the conclusions shall have been based shall have been performed not earlier than sixty (60) days prior to the date of such report.
- 4.2.3 Post-Term Access. If and to the extent reasonably practical, Tenant shall use commercially reasonable efforts to complete (i) any environmental testing prior to the expiration of the Term or as soon thereafter as is reasonably practical and (ii) any remediation, if applicable, as soon after the Term as is reasonably practical, in each case taking into account, among other things, the conditions at the Properties and the requirements of any relevant governmental authority. If completion of such testing and, if applicable, remediation, shall not have been completed prior to the expiration of the Term, then, following the expiration or earlier termination of the Term, Tenant and its agents, employees and environmental consultants shall have reasonable rent-free access to each applicable Property, upon reasonable advanced notice to Landlord, for the sole purposes of (i) conducting the review and assessment and related remediation, if applicable, necessary to prepare and deliver the Environmental Reports as contemplated by Section 4.2.2 and/or (ii) performing Tenant's obligations, if any, pursuant to Section 4.2.1 in respect of any Leased Property to the extent the same were not completed during the Term. Tenant shall exercise such access rights in a manner designed to minimize, to the extent reasonably practical, any material interference with site operations then being conducted at the Property. Tenant shall indemnify and hold Landlord harmless from any claims, liabilities and reasonable out-of-pocket costs and expenses incurred by Landlord as a result of Tenant's (or its agents', employees' or environmental consultants') activities at the applicable Properties taken in connection with Tenant's post-term access rights pursuant to this Section 4.2.3, provided that Tenant shall have the right to control the defense of any such claim. The foregoing indemnity shall not extend to any claims or liabilities resulting from the gross negligence or willful misconduct of Landlord of any Affiliated Person of Landlord or any Person acting on behalf of or with the consent of any of the foregoing.

- 4.2.4 <u>Underground Storage Tanks</u>. It is expressly understood and agreed that Tenant's obligations under this Agreement shall include the maintenance and, if necessary, replacement of underground storage tanks at the Leased Property.
- 4.2.5 <u>Survival</u>. The provisions of <u>Sections 4.2.1</u>, <u>4.2.2</u> and <u>4.2.3</u> shall survive the expiration or sooner termination of this Agreement.

4.3 Ground Leases.

- 4.3.1 <u>Tenant's Obligations</u>. Tenant shall pay and perform all of Landlord's obligations as tenant under the Ground Leases.
- 4.3.2 <u>Landlord's Obligations</u>. Landlord shall provide Tenant copies of all notices received by Landlord from the lessor under any Ground Lease, promptly upon receipt thereof, unless it is apparent from any such notice that the lessor also provided a copy thereof to Tenant directly. Landlord shall not amend, modify or supplement any Ground Lease or enter into any new ground lease (including, for certainty, any ground or master lease with respect to any Property or portion thereof) without Tenant's consent, and Landlord shall not take or permit any others acting on its behalf (but, for certainty, specifically excluding Tenant) to take any action constituting or resulting in a default under any Ground Lease.
- 4.3.3 Options. If Landlord has the right, under the provisions of any of the Ground Leases, to elect to renew or extend the term of any Ground Leases or to exercise a purchase option, right of first offer or right of first refusal with respect to the property demised thereby, Tenant shall so notify Landlord, in the case of renewal or extension, at least one hundred eighty (180) days (but no more than one (1) year) prior to the expiration of the period within which Landlord is obligated to notify the lessor under such Ground Leases of its election to renew or extend, as the case may be, and, in the case of a purchase option, right of first offer or right of first refusal, promptly upon Tenant receiving notice from the landlord under such Ground Lease of Landlord's right to make an election with respect to the purchase of such property. Such notice from Tenant shall contain (1) all of the relevant facts about the impending election to renew, extend or purchase, including, as applicable, the length of the period of renewal, the rental rate and/or the purchase price and (2) Tenant's election as to whether or not Tenant wishes to exercise such election or purchase rights, as the case may be. If Tenant desires that Landlord exercise such election or purchase rights, Landlord and Tenant shall cooperate as necessary to enable and effect, in the case of a renewal or extension, Landlord's exercise thereof, and, in the case of a purchase option, right of first offer or right of first refusal, the exercise of such option for Tenant (or Tenant's designee) to purchase the Property. If Tenant instead notices Landlord of its desire not to extend or renew such Ground Lease beyond the then-current term, then (whether or not Landlord exercises such extension or renewal right) this Agreement shall terminate with respect to such Property and Ground Lease at the expiration of the then-current term of such Ground Lease; provided, however, in such event, there shall be no reduction in the Minimum Rent.
- 4.4 <u>Shell Agreement</u>. Tenant acknowledges its obligations under the Shell Agreement, and Landlord and Tenant agree that this Agreement shall, for purposes of Section 2 of the Subordination, Non-Disturbance and Attornment Agreement among Landlord, HPT PSC

Properties Trust, HPT PSC Properties LLC, Tenant and Shell entered into as of April 15, 2013 ("Shell SNDA") in connection with the Shell Agreement, constitute a replacement "Lease", as defined in the Shell SNDA, for the Prior Lease. Tenant shall indemnify, defend and hold Landlord harmless from and against any actual loss, damages, claims and liabilities and reasonable out-of-pocket costs and expenses incurred by Landlord as a result of any claim of a breach by Landlord under the Shell Agreement to the extent attributable to the actions of Tenant or any Affiliated Person of Tenant (and, for certainty, not of Landlord or any Affiliated Person of Landlord) during the Term, provided that Tenant shall have the right to control the defense of any such claim.

ARTICLE 5 MAINTENANCE AND REPAIRS

5.1 Maintenance and Repair.

- 5.1.1 Tenant's Obligations. Tenant shall keep (or cause to be kept), at Tenant's sole cost and expense, the Leased Property and all private roadways, sidewalks and curbs appurtenant thereto (and Tenant's Personal Property) in good order and repair, reasonable wear and tear excepted (whether or not the need for such repairs occurs as a result of Tenant's use, any prior use, the elements or the age of the Leased Property or Tenant's Personal Property or any portion thereof), and shall promptly make or cause to be made all necessary and appropriate repairs and replacements thereto of every kind and nature, whether interior or exterior, structural or nonstructural, ordinary or extraordinary, foreseen or unforeseen or arising by reason of a condition existing prior to the commencement of the Term (concealed or otherwise). All repairs shall be made in a good, workmanlike manner, consistent with industry standards for comparable travel centers in like locales, in accordance with all applicable federal, state and local statutes, ordinances, codes, rules and regulations relating to any such work. Tenant shall not take or omit to take (or permit any Person to take or omit to take) any action, the taking or omission of which would materially and adversely impair the value or the usefulness of the Leased Property or any material part thereof for its Permitted Use. Tenant's obligations under this Section 5.1.1 shall be limited in the event of any casualty or Condemnation as set forth in Article 10 and Article 11 and Tenant's obligations with respect to Hazardous Substances are as set forth in Section 4.2.
- 5.1.2 <u>Landlord's Obligations</u>. Landlord shall not, under any circumstances, be required to build or rebuild any improvement on the Real Property, or to make any repairs, replacements, alterations, restorations or renewals of any nature or description to the Leased Property, whether ordinary or extraordinary, structural or nonstructural, foreseen or unforeseen, or to make any expenditure whatsoever with respect thereto, or to maintain the Leased Property in any way. Except as otherwise expressly provided in this Agreement, Tenant hereby waives, to the maximum extent permitted by law, the right to make repairs at the expense of Landlord pursuant to any law. Landlord shall have the right to give, record and post, as appropriate, notices of nonresponsibility under any mechanic's lien laws now or hereafter existing.
- 5.1.3 Nonresponsibility of Landlord, Etc. All materialmen, contractors, artisans, mechanics and laborers and other persons contracting with Tenant with respect to the Leased Property, or any part thereof, are hereby charged with notice that liens on the Leased

Salem. OR

Property or on Landlord's interest therein are expressly prohibited and that they must look solely to Tenant to secure payment for any work done or material furnished to Tenant or any Manager

or for any other purpose during the term of this Agreement.

Nothing contained in this Agreement shall be deemed or construed in any way as constituting the consent or request of Landlord, express or implied, by inference or otherwise, to any contractor, subcontractor, laborer or materialmen for the performance of any labor or the furnishing of any materials for any alteration, addition, improvement or repair to the Leased Property or any part thereof or as giving Tenant any right, power or authority to contract for or permit the rendering of any services or the furnishing of any materials that would give rise to the filing of any lien against the Leased Property or any part thereof nor to subject Landlord's estate in the Leased Property or any part thereof to liability under any mechanic's lien law of any State in any way, it being expressly understood Landlord's estate shall not be subject to any such liability.

- 5.2 <u>Tenant's Personal Property</u>. Tenant shall provide and maintain (or cause to be provided and maintained) throughout the Term all such Tenant's Personal Property as shall be necessary in order to operate in compliance with applicable material Legal Requirements and Insurance Requirements and otherwise in accordance with customary practice in the industry for the Permitted Use.
- 5.3 <u>Yield Up</u>. Upon the expiration or sooner termination of this Agreement, Tenant shall remove all of Tenant's Personal Property (other than that purchased by Landlord pursuant to <u>Article 15</u>) and vacate and surrender the Leased Property to Landlord in substantially the same condition in which the Leased Property was in on the Effective Date, except as repaired, rebuilt, restored, altered or added to as permitted or required by the provisions of this Agreement, reasonable wear and tear excepted (and casualty damage and Condemnation, in the event that this Agreement is terminated following a casualty or Condemnation in accordance with <u>Article 10</u> or <u>Article 11</u>, excepted).

In addition, upon the expiration or earlier termination of this Agreement, Tenant shall, at Landlord's sole cost and expense, use its good faith efforts to transfer (or cause to be transferred) to Landlord or its nominee, and cooperate with Landlord or Landlord's nominee in connection with the processing of all applications for, licenses, operating permits and other governmental authorizations and all contracts, including contracts with Government Agencies and rights with third party franchisors which may be necessary for the use and operation of the Travel Centers as then operated (all such licenses, permits, authorizations and contracts, but excluding any trademarks, tradenames and other intellectual property, being "Operating Rights"). Tenant hereby appoints Landlord as its attorney-in-fact, with full power of substitution, for the purpose of carrying out the provisions of this paragraph and taking any action, including, without limitation, executing, delivering and filing applications, certificates, instruments and other documents and papers with Government Agencies, and executing any instruments, assignments, conveyances, and other transfers which are required to be taken or executed by Tenant, on its behalf and in its name, which appointment is coupled with an interest, is irrevocable and durable and shall survive the subsequent dissolution of Tenant.

If requested by Landlord, Tenant shall continue to manage one or more of the Travel Centers for Landlord (as a third-party manager) after the expiration of the Term for up to one hundred eighty (180) days, on such customary arms'-length terms (including receipt by Tenant of a market management fee), as may be agreed by the parties, acting reasonably.

ARTICLE 6 IMPROVEMENTS, ETC.

Tenant may make, construct or install (or permit to be made, constructed or installed) any Capital Additions provided that construction or installation of the same will not adversely affect or violate any material Legal Requirement or Insurance Requirement applicable to any Property and shall not adversely affect the market value of the applicable Property. No Capital Addition shall be made which would tie in or connect any Leased Improvements with any other improvements on property adjacent to any Property (and not part of the Land) including, without limitation, tie-ins of buildings or other structures or utilities. Any improvements shall, upon the expiration or sooner termination of this Agreement, remain or pass to and become the property of Landlord, free and clear of all encumbrances other than Permitted Encumbrances.

ARTICLE 7 LIENS

Subject to Article 8, Tenant shall use its best efforts not, directly or indirectly, to create or allow to remain and shall promptly discharge (or cause to be discharged), at its expense, any lien, encumbrance, attachment, title retention agreement or claim upon the Leased Property, or any portion thereof, or Tenant's leasehold interest therein or any attachment, levy, claim or encumbrance in respect of the Rent, other than (a) Permitted Encumbrances, (b) restrictions, liens and other encumbrances which are consented to in writing by Landlord, (c) liens for those taxes of Landlord which Tenant is not required to pay hereunder, (d) subleases permitted by Article 16, (e) liens for Impositions or for sums resulting from noncompliance with Legal Requirements so long as (i) the same are not yet due and payable, or (ii) are being contested in accordance with Article 8, (f) liens of mechanics, laborers, materialmen, suppliers or vendors incurred in the ordinary course of business that are not yet due and payable or are for sums that are being contested in accordance with Article 8, (g) any Property Mortgages or other liens which are the responsibility of Landlord pursuant to the provisions of Article 20 and (h) Landlord Liens and any other voluntary liens created by Landlord.

ARTICLE 8 PERMITTED CONTESTS

Tenant shall have the right to contest the amount or validity of any Imposition, Legal Requirement, Insurance Requirement, Environmental Obligation, lien, attachment, levy, encumbrance, charge or claim (collectively, "Claims") as to the Leased Property, by appropriate legal proceedings, conducted in good faith and with due diligence, provided that (a) the foregoing shall in no way be construed as relieving, modifying or extending Tenant's obligation to pay (or cause to be paid) any Claims as finally determined, (b) such contest shall not cause Landlord or Tenant to be in default under any Ground Lease, mortgage or deed of trust encumbering the Leased Property, or any portion thereof (Landlord agreeing that any such

Ground Lease, mortgage or deed of trust shall permit Tenant to exercise the rights granted pursuant to this Article 8) or any interest therein or result in or reasonably be expected to result in a lien attaching to the Leased Property, or any portion thereof, (c) no part of the Leased Property nor any Rent therefrom shall be in any immediate danger of sale, forfeiture, attachment or loss, and (d) Tenant shall indemnify and hold harmless Landlord from and against any cost, claim, damage, penalty or reasonable expense, including reasonable attorneys' fees, incurred by Landlord in connection therewith or as a result thereof, provided that Tenant shall have the right to control the defense of any such claim. Landlord agrees to join in any such proceedings if required legally to prosecute such contest, provided that Landlord shall not thereby be subjected to any liability therefor (including, without limitation, for the payment of any costs or expenses in connection therewith) unless Tenant agrees by agreement in form and substance reasonably satisfactory to Landlord, to assume and indemnify Landlord with respect to the same. Tenant shall be entitled to any refund of any Claims and such charges and penalties or interest thereon which have been paid by Tenant or paid by Landlord to the extent that Landlord has been fully reimbursed by Tenant. If Tenant shall fail (x) to pay or cause to be paid any Claims when finally determined, (y) to provide reasonable security therefor or (z) to prosecute or cause to be prosecuted any such contest diligently and in good faith, Landlord may, upon reasonable notice to Tenant (which notice shall not be required if Landlord shall reasonably determine that the same is not practicable), pay such charges, together with interest and penalties due with respect thereto, and Tenant shall reimburse Landlord therefor, upon demand, as Additional Charges.

ARTICLE 9 INSURANCE AND INDEMNIFICATION

- 9.1 General Insurance Requirements. Tenant shall, at all times during the Term and at any other time Tenant shall be in possession of any Property, or any portion thereof, keep (or cause to be kept) such Property and all property located therein or thereon, insured against the risks and in such amounts as required by Landlord, acting reasonably, provided in each case that such insurance coverage is available on commercially reasonable terms.
- 9.2 Waiver of Subrogation. Landlord and Tenant agree that (insofar as and to the extent that such agreement may be effective without invalidating or making it impossible to secure insurance coverage from responsible insurance companies doing business in any State) with respect to any property loss which is covered by insurance then being carried by Landlord or Tenant, the party carrying such insurance and suffering said loss releases the others of and from any and all claims with respect to such loss; and they further agree that their respective insurance companies (and, if Landlord or Tenant shall self insure in accordance with the terms hereof, Landlord or Tenant, as the case may be) shall have no right of subrogation against the other on account thereof, even though extra premium may result therefrom. In the event that any extra premium is payable by Tenant as a result of this provision, Landlord shall not be liable for reimbursement to Tenant for such extra premium.
- 9.3 Form Satisfactory, Etc. All insurance policies and endorsements required pursuant to this Article 9 shall be fully paid for, nonassessable, and issued by reputable insurance companies authorized to do business in the State and having a general policy holder's rating of no less than A- in Best's latest rating guide. At all times, all property, business interruption, liability and flood insurance policies, with the exception of worker's compensation and

SEP 0 2 2025 Salem, OR

employer's liability insurance coverage, shall include Landlord and any Property Mortgagee as additional insureds or loss payees, as their interests may appear. All loss adjustments shall be payable as provided in <u>Article 10</u>, except that losses under liability and worker's compensation insurance policies shall be payable directly to the party entitled thereto. Tenant shall cause all insurance premiums to be paid and shall deliver (or cause to be delivered) certificates thereof to Landlord as soon as possible after their effective date. All such policies shall provide Landlord (and any Property Mortgagee if required by the same) thirty (30) days prior written notice of any material change or cancellation of such policy, subject to policy terms and conditions.

- 9.4 <u>Self-Insurance</u>. Notwithstanding the foregoing and where permitted by Law, Tenant shall have the right not to maintain insurance as stated above by providing Landlord at the outset (or, if self-insurance constitutes a change, at least thirty (30) days' prior written of such change) of Tenant's election to self-insure the same, so long as Tenant is an Affiliated Person of BP Parent or Guarantor is a Qualifying Guarantor. With respect to such self-insurance and to the extent of Tenant's express obligations assumed in this Agreement, Tenant hereby waives and releases Landlord from any and all claims, losses, expenses, damages and liability for which Landlord is or may be held liable based on or arising out of any act, occurrence or inaction that would have been covered by such insurance had Tenant maintained the same.
- 9.5 Indemnification of Landlord. Notwithstanding the existence of any insurance provided for herein and without regard to the policy limits of any such insurance, Tenant shall protect, indemnify and hold harmless Landlord for, from and against all liabilities, obligations, claims, damages, penalties, causes of action, costs and reasonable expenses (including, without limitation, reasonable attorneys' fees), to the maximum extent permitted by law, imposed upon or incurred by or asserted against Landlord by reason of the following, except to the extent caused by the gross negligence or willful misconduct of Landlord or any Person acting on Landlord's behalf, and provided that Tenant shall have the right to control the defense of any such claim: (a) any accident or injury to, or death of, persons or loss of or damage to property occurring on or about any Property or portion thereof or adjoining sidewalks or rights of way during the Term, (b) any past, present or future condition or use, misuse, non-use, management, maintenance or repair by Tenant, any Manager or anyone claiming under any of them of any Property, Tenant's Personal Property or any litigation, proceeding or claim by governmental entities (other than Condemnation proceedings), or other third parties relating to any Property or portion thereof or Tenant's Personal Property or such use, misuse, non-use, condition, management, maintenance, or repair thereof, including failure to perform obligations under this Agreement, to which Landlord is made a party during the Term (limited, in the case of Environmental Obligations, to those provided in Section 4.2.1), (c) any Impositions that are the obligations of Tenant to pay pursuant to the applicable provisions of this Agreement, (d) any failure on the part of Tenant or anyone claiming under Tenant to perform or comply with any of the terms of this Agreement and/or (e) any of the assets owned or businesses conducted by Tenant or Person Controlling, Controlled by or under common Control with Tenant, whenever arising. Tenant, at its expense, shall contest, resist and defend any such claim, action or proceeding asserted or instituted against Landlord (and shall not be responsible for any duplicative attorneys' fees incurred by Landlord) or may compromise or otherwise dispose of the same, with Landlord's prior written consent (which consent may not be unreasonably withheld, delayed or conditioned). The obligations of Tenant under this Section 9.5 shall survive the termination of this Agreement.

ARTICLE 10 CASUALTY

10.1 Insurance Proceeds. Except as provided in the last clause of this sentence, all proceeds payable by reason of any loss or damage to any Property, or any portion thereof, and insured under any policy of insurance (except any self-insurance or insurance obtained directly or indirectly from captive insurers which are Affiliated Persons of Tenant) required by Article 9 (other than the proceeds of any business interruption insurance or insurance proceeds for Tenant's Personal Property) shall be paid directly to Landlord (subject to the provisions of Section 10.2) and all loss adjustments with respect to losses payable to Landlord shall require the prior written consent of Landlord, which consent shall not be unreasonably withheld, delayed or conditioned; provided, however, that, so long as no Event of Default shall have occurred and be continuing, all such proceeds less than or equal to Ten Million Dollars (\$10,000,000) shall be paid directly to Tenant and such losses may be adjusted without Landlord's consent. If Tenant is required to reconstruct or repair any Property as provided herein, such proceeds shall be paid out by Landlord from time to time for the reasonable costs of reconstruction or repair of such Property necessitated by such damage or destruction, subject to and in accordance with the provisions of Section 10.2.3. Any excess proceeds of insurance remaining after the completion of the restoration shall be paid to Tenant. In the event that the provisions of Section 10.2.1 are applicable, the insurance proceeds shall be retained by the party entitled thereto pursuant to Section 10.2.1. In the event that Tenant relies on self-insurance or insurance obtained directly or indirectly from captive insurers which are Affiliated Persons of Tenant for certain coverages, and a casualty loss is sustained, Tenant shall fund the loss to the extent of Tenant's express obligations under this Agreement.

10.2 Damage or Destruction.

10.2.1 Termination Due to Damage or Destruction of Leased Property. If, during the Term, any Property shall be totally or partially destroyed and the Travel Center located thereon is thereby rendered Unsuitable for Its Permitted Use, Tenant may, by the giving of Notice thereof to Landlord, terminate this Agreement with respect to such affected Property, whereupon, this Agreement shall terminate with respect to such affected Property, and Tenant shall, prior to such termination, pay to Landlord (i) if covered by independent third party insurance, the amount of any unpaid deductible under the applicable insurance policies covering such Travel Center and the amount of any uninsured loss and any difference between the replacement cost of the affected Property and the casualty insurance proceeds therefor (if and to the extent such uninsured loss or difference is the result of Tenant's failure to maintain insurance coverages as required hereunder), and (ii) if not covered by third party insurance maintained in compliance with this Agreement, the amount of the replacement cost of the affected Property, whereupon from and after such termination the Minimum Rent shall be reduced by an annual amount equal to the product of (a) the Percentage Reduction multiplied by (b) the total amount received by Landlord on account of such casualty.

10.2.2 Other Damage or Destruction. If, during the Term, any Property shall be totally or partially destroyed but the Travel Center located thereon is not rendered Unsuitable for Its Permitted Use or Tenant does not elect to terminate this Agreement with respect to the

affected Property, then Tenant shall promptly restore such Travel Center as provided in Salem, OR Section 10.2.3.

produce of the

- Property pursuant to Section 10.2 and this Agreement is not terminated as to such Property pursuant to this Article 10, Tenant shall commence (or cause to be commenced) promptly and continue diligently to perform (or cause to be performed) the repair and restoration of such Property (hereinafter called the "Work"), so as to restore (or cause to be restored) the applicable Property in material compliance with all Legal Requirements and so that such Property shall be, to the extent practicable, at least substantially equivalent in value and general utility to its general utility and value immediately prior to such damage or destruction.
- 10.3 <u>Damage Near End of Term</u>. Notwithstanding any provisions of <u>Section 10.1</u> or <u>10.2</u> to the contrary, if damage to or destruction of any Property occurs during the last twelve (12) months of the Term and if such damage or destruction cannot reasonably be expected to be fully repaired and restored prior to the date that is six (6) months prior to the end of the Term, the provisions of <u>Section 10.2.1</u> shall apply as if such Property had been totally or partially destroyed and the Travel Center thereon rendered Unsuitable for Its Permitted Use.
- 10.4 Restoration of Tenant's Personal Property. If Tenant is required to restore any Property as hereinabove provided, Tenant shall either (a) restore all alterations and improvements made by Tenant and Tenant's Personal Property, as the case may be, or (b) replace such alterations and improvements and Tenant's Personal Property with improvements or items of the same or better quality and utility in the operation of such Property.
- 10.5 No Abatement of Rent. Other than as specifically provided in this Agreement, Tenant's obligation to make all payments of Rent and to pay all other charges as and when required under this Agreement shall remain unabated during the Term notwithstanding any damage involving the Leased Property, or any portion thereof. The provisions of this Article 10 shall be considered an express agreement governing any cause of damage or destruction to the Leased Property, or any portion thereof, to the maximum extent permitted by law.
- 10.6 <u>Waiver</u>. Tenant hereby waives any statutory rights of termination which may arise by reason of any damage or destruction of the Leased Property, or any portion thereof.

ARTICLE 11 CONDEMNATION

- 11.1 <u>Total Condemnation, Etc.</u> If either (i) the whole of any Property shall be taken by Condemnation or (ii) a Condemnation of less than the whole of any Property renders any Property Unsuitable for Its Permitted Use, this Agreement shall terminate with respect to such Property, and Tenant and Landlord shall seek the Award for their interests in the applicable Property as provided in <u>Section 11.5</u>. Upon payment to Landlord of any such Award, the Minimum Rent shall thereafter be reduced by an annual amount equal to the product of (a) the Percentage Reduction multiplied by (b) the amount of such Award received by Landlord.
- 11.2 <u>Partial Condemnation</u>. In the event of a Condemnation of less than the whole of any Property such that such Property is still suitable for its Permitted Use, Tenant shall,

regardless of the extent of the Award, commence (or cause to be commenced) promptly and continue diligently to restore (or cause to be restored) the untaken portion of the applicable Leased Improvements so that such Leased Improvements shall constitute a complete architectural unit of the same general character and condition (as nearly as may be possible under the circumstances) as such Leased Improvements existing immediately prior to such Condemnation, in material compliance with all Legal Requirements, subject to the provisions of this Section 11.2.

Subject to the terms hereof and after Tenant has funded any deficiency in the amount of the Award received by Landlord to complete such restoration, Landlord shall contribute to the cost of restoration that part of the Award received by Landlord and necessary to complete such restoration, together with severance and other damages awarded to Landlord for the taken Leased Improvements, to Tenant regularly during the restoration period so as to permit payment for the cost of such repair or restoration. Landlord may, at its option, reasonably condition advancement of such portion of the Award and other amounts on (a) its approval of plans and specifications of an architect satisfactory to Landlord (which approval shall not be unreasonably withheld, delayed or conditioned), (b) general contractors' estimates, (c) architect's certificates, (d) conditional lien waivers of general contractors, if available, (e) evidence of approval by all governmental authorities and other regulatory bodies whose approval is required, and (f) such other certificates as Landlord may, from time to time, reasonably require. Landlord's obligation under this Section 11.2 to disburse the Award and such other amounts shall be subject to (x) the collection thereof by Landlord and (y) the satisfaction of any applicable requirements of any Property Mortgage, and the release of such Award by the applicable Property Mortgagee. Tenant's obligation to restore the Leased Property pursuant to this Article 11 shall be subject to the release of any portion of the Award by Landlord (as provided above) and, if applicable, by the applicable Property Mortgagee to Landlord or directly to Tenant.

- Agreement of Rent. Other than as specifically provided in this Agreement, this Agreement shall remain in full force and effect and Tenant's obligation to make all payments of Rent and to pay all other charges as and when required under this Agreement shall remain unabated during the Term notwithstanding any Condemnation involving the Leased Property, or any portion thereof. The provisions of this Article 11 shall be considered an express agreement governing any Condemnation involving the Leased Property and, to the maximum extent permitted by law, no local or State statute, law, rule, regulation or ordinance in effect during the Term which provides for such a contingency shall have any application in such case.
- Property or Tenant's interest therein, this Agreement shall continue in full force and effect and Tenant shall continue to pay (or cause to be paid), in the manner and on the terms herein specified, the full amount of the Rent. Tenant shall continue to perform and observe (or cause to be performed and observed) all of the other terms and conditions of this Agreement on the part of the Tenant to be performed and observed. The entire amount of any Award made for such temporary Condemnation allocable to the Term, whether paid by way of damages, rent or otherwise, shall be paid to Tenant. Tenant shall, promptly upon the termination of any such period of temporary Condemnation, at its sole cost and expense, restore the affected Property to the condition that existed immediately prior to such Condemnation, in material compliance with all applicable Legal Requirements, unless such period of temporary Condemnation shall extend

beyond the expiration of the Term, in which event Tenant shall not be required to make such restoration.

11.5 <u>Allocation of Award</u>. Except as provided in <u>Section 11.2</u> and <u>11.4</u> and the second sentence of this <u>Section 11.5</u>, the total Award shall be solely the property of and payable to Landlord. Any portion of the Award made for the taking of Tenant's leasehold interest in the Leased Property, loss of business during the remainder of the Term, the taking of Tenant's Personal Property, the taking of Capital Additions paid for by Tenant and Tenant's removal and relocation expenses shall be the sole property of and payable to Tenant. In any Condemnation proceedings, Landlord and Tenant shall each seek its own Award in conformity herewith, at its own expense.

ARTICLE 12 DEFAULTS AND REMEDIES

- 12.1 <u>Events of Default</u>. The occurrence of any one or more of the following events shall constitute an "Event of Default" hereunder:
 - (a) should Tenant fail to make any payment of the Rent or any other sum payable hereunder when due and should such failure continue for a period of five (5) Business Days after Notice thereof from Landlord to Tenant; or
 - (b) should Tenant default in the due observance or performance of any of the terms, covenants or agreements contained herein to be performed or observed by it (other than as specified in clause (a) above) and should such default continue for a period of thirty (30) days after Notice thereof from Landlord to Tenant; provided, however, that if such default is susceptible of cure but such cure cannot be accomplished with due diligence within such period of time and if, in addition, Tenant commences to cure or cause to be cured such default within thirty (30) days after Notice thereof from Landlord and thereafter prosecutes the curing of such default with all due diligence, such period of time shall be extended to such period of time (not to exceed an additional one hundred and fifty (150) days in the aggregate) as may be necessary to cure such default with all due diligence; or
 - (c) should any default occur and be continuing under any Guaranty beyond applicable notice and cure periods provided below; or
 - (d) should Tenant or any Guarantor generally not be paying its debts as they become due or should Tenant or any Guarantor make a general assignment for the benefit of creditors; or
 - (e) should any petition be filed by or against Tenant or any Guarantor under the Federal bankruptcy laws, or should any other proceeding be instituted by or against Tenant or any Guarantor seeking to adjudicate Tenant or any Guarantor a bankrupt or insolvent, or seeking liquidation, reorganization, arrangement, adjustment or composition of Tenant's or any Guarantor's debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors, or seeking the entry of an order for relief or the appointment of a receiver, trustee, custodian or other similar official for Tenant or

Guarantor or for any substantial part of the property of Tenant or any Guarantor and such proceeding is not dismissed within one hundred eighty (180) days after institution thereof; or

- (f) should Tenant or any Guarantor cause or institute any proceeding for its dissolution or termination; or
- (g) should the estate or interest of Tenant in the Leased Property or any part thereof be levied upon or attached in any proceeding and the same shall not be vacated or discharged within the later of (x) ninety (90) days after commencement thereof, unless the amount in dispute is less than \$250,000, in which case Tenant shall give Notice to Landlord of the dispute but Tenant may defend in any suitable way, and (y) two hundred seventy (270) days after receipt by Tenant of Notice thereof from Landlord (unless Tenant shall be contesting such lien or attachment in good faith in accordance with Article 8),

then, and in any such event, Landlord, in addition to all other remedies available to it, may terminate this Agreement with respect to any or all of the Leased Property by giving at least fifteen (15) Business Days' Notice thereof to Tenant and upon the expiration of the time, if any, fixed in such Notice, this Agreement shall terminate with respect to all or the designated portion of the Leased Property and all rights of Tenant under this Agreement with respect thereto shall cease. Landlord shall have and may exercise all rights and remedies available at law and in equity to Landlord as a result of Tenant's breach of this Agreement.

Upon the termination of this Agreement in connection with any Event of Default, Landlord may, in addition to any other remedies provided herein (including the rights set forth in Section 5.3), enter upon the Real Property, or any portion thereof and take possession thereof, without liability for trespass or conversion (Tenant hereby waiving any right to notice or hearing prior to such taking of possession by Landlord).

Notwithstanding the foregoing, Tenant shall be entitled, at any time following the occurrence of an Event of Default with respect to Guarantor under Sections 12.1(c), 12.1(d), 12.1(e) or 12.1(f), to cure the Event of Default by causing another Person who meets the requirements of a Qualifying Guarantor to execute and deliver to Landlord a Guaranty or joinder to the existing Guaranty pursuant to which that other Person, at Tenant's option, either (a) shall execute and deliver to Landlord a new Guaranty (in which case, the existing Guarantor shall automatically be released from any obligations under its Guaranty that are obligations of the Qualifying Guarantor under such new Guaranty or otherwise first arise or accrue after the date of such new Guaranty (it being understood that in no circumstance will the existing Guarantor be deemed liable for any obligations that otherwise first arise or accrue under this Agreement from and after the date of such new Guaranty)) or (b) shall become jointly and severally liable with the existing Guarantor under the existing Guaranty and, in any such case, all references herein to "Guarantor" will be considered to be to the new Guarantor in lieu of the existing Guarantor or, if applicable, to include the Person joining with the existing Guarantor. The delivery of such undertaking by a new Qualifying Guarantor as described above within fifteen (15) Business Days (time being of the essence) of Tenant's receipt of the Notice of an Event of Default with respect to the existing Guarantor under Sections 12.1(c), 12.1(d), 12.1(e) or 12.1(f) shall operate

to cure such Event of Default, and this Agreement shall continue in full force and effect. OR notwithstanding that the condition involving such existing Guarantor continues to exist (and that continuing condition will no longer constitute an Event of Default).

Remedies. None of the termination of this Agreement pursuant to Section 12.1, nor, in each case following such termination, (a) the repossession of the Leased Property, or any portion thereof, (b) the failure of Landlord to relet the Leased Property, or any portion thereof, or (c) the reletting of all or any of portion of the Leased Property, in any case, shall relieve Tenant of its liability and obligations hereunder, all of which shall survive any such termination, repossession or reletting. In the event of any such termination, Tenant shall forthwith pay to Landlord all Rent due and payable with respect to the Leased Property, or terminated portion thereof, through and including the date of such termination. Thereafter, Tenant, until the end of what would have been the Term of this Agreement in the absence of such termination, and whether or not the Leased Property, or any portion thereof, shall have been relet, shall be liable to Landlord for, and shall pay to Landlord, as current damages, the Rent (Additional Charges to be reasonably calculated by Landlord) which would be payable hereunder for the remainder of the Term had such termination not occurred, less the net proceeds, if any, of any reletting of the Leased Property, or any portion thereof, after deducting all reasonable expenses in connection with such reletting, including, without limitation, all repossession costs, brokerage commissions, legal expenses, attorneys' fees, advertising, expenses of employees, alteration costs and expenses of preparation for such reletting. Tenant shall pay such current damages to Landlord monthly on the days on which the Minimum Rent would have been payable hereunder if this Agreement had not been so terminated with respect to such of the Leased Property.

At any time after such termination, whether or not Landlord shall have collected any such current damages, as liquidated final damages beyond the date of such termination, at Landlord's election, Tenant shall pay to Landlord an amount equal to the present value (as reasonably determined by Landlord using a discount rate equal to five percent (5%) per annum) of the excess, if any, of the Rent and other charges which would be payable hereunder from the date of such termination (assuming that, for the purposes of this paragraph, annual payments by Tenant on account of Additional Charges would be the same as payments required for the immediately preceding twelve calendar months, or if less than twelve calendar months have expired since the Effective Date, the payments required for such lesser period projected to an annual amount) for what would be the then unexpired term of this Agreement if the same remained in effect, over the fair market rental for the same period. Nothing contained in this Agreement shall, however, limit or prejudice the right of Landlord to prove and obtain in proceedings for bankruptcy or insolvency an amount equal to the maximum allowed by any statute or rule of law in effect at the time when, and governing the proceedings in which, the damages are to be proved, whether or not the amount be greater than, equal to, or less than the amount of the loss or damages referred to above.

In case of any Event of Default, re-entry, expiration and dispossession by summary proceedings or otherwise, Landlord may, (a) relet the Leased Property or any part or parts thereof, either in the name of Landlord or otherwise, for a term or terms which may at Landlord's option, be equal to, less than or exceed the period which would otherwise have constituted the balance of the Term and may grant concessions or free rent to the extent that Landlord considers advisable and necessary to relet the same, and (b) may make such reasonable

alterations, repairs and decorations in the Leased Property, or any portion thereof, as Landlord, in its sole and absolute discretion, considers advisable and necessary for the purpose of reletting the Leased Property; and the making of such alterations, repairs and decorations shall not operate or be construed to release Tenant from liability hereunder as aforesaid. Landlord shall in no event be liable in any way whatsoever for any failure to relet all or any portion of the Leased Property, or, in the event that the Leased Property is relet, for failure to collect the rent under such reletting. To the maximum extent permitted by law, Tenant hereby expressly waives any and all rights of redemption granted under any present or future laws in the event of Tenant being evicted or dispossessed, or in the event of Landlord obtaining possession of the Leased Property, by reason of the occurrence and continuation of an Event of Default hereunder.

Notwithstanding anything to the contrary set forth in this Agreement, if an Event of Default shall be triggered solely with respect to any of Sections 9.5(d), 12.1(c), 21.1, 21.2, 21.4 or 21.5 (and not with respect to any other Section of this Agreement), in no event shall the damages recovered by Landlord pursuant to this Agreement exceed an amount equal to the sum of (i) the present value (as reasonably determined by Landlord using a discount rate equal to ten and eight tenths percent (10.8%) per annum) of the Minimum Rent which would be payable hereunder from the date of such termination for what would be the then unexpired Term of this Agreement if the same remained in effect; and (ii) all amounts due and unpaid under this Agreement as of the date of the occurrence of the Event of Default.

- 12.3 <u>Tenant's Waiver</u>. IF THIS AGREEMENT IS TERMINATED PURSUANT TO <u>SECTIONS 12.1</u> OR <u>12.2</u>, TENANT WAIVES, TO THE EXTENT PERMITTED BY LAW, ANY RIGHT TO A TRIAL BY JURY IN THE EVENT OF SUMMARY PROCEEDINGS TO ENFORCE THE REMEDIES SET FORTH IN THIS <u>ARTICLE 12</u>, AND THE BENEFIT OF ANY LAWS NOW OR HEREAFTER IN FORCE EXEMPTING PROPERTY FROM LIABILITY FOR RENT OR FOR DEBT.
- 12.4 <u>Application of Funds</u>. Any payments received by Landlord under any of the provisions of this Agreement during the existence or continuance of any Event of Default (and any payment made to Landlord rather than Tenant due to the existence of any Event of Default) shall be applied to Tenant's current and past due obligations under this Agreement in such order as Landlord may determine or as may be prescribed by the laws of the State. Any balance shall be paid to Tenant.
- 12.5 Landlord's Right to Cure Tenant's Default. If an Event of Default shall have occurred and be continuing, Landlord, after Notice to Tenant (which Notice shall not be required if Landlord shall reasonably determine immediate action is necessary to protect person or property), without waiving or releasing any obligation of Tenant and without waiving or releasing any Event of Default, may (but shall not be obligated to), at any time thereafter, make such payment or perform such act for the account and at the expense of Tenant, and may, to the maximum extent permitted by law, enter upon the Real Property, or any portion thereof, for such purpose and take all such action thereon as, in Landlord's sole and absolute discretion, may be necessary or appropriate therefor. No such entry shall be deemed an eviction of Tenant. All reasonable costs and expenses (including, without limitation, reasonable attorneys' fees) incurred by Landlord in connection therewith, together with interest thereon (to the extent permitted by

law) at the Overdue Rate from the date such sums are paid by Landlord until repaid, shall be paid by Tenant to Landlord, on demand.

ARTICLE 13 HOLDING OVER

Any failure by Tenant to surrender possession of any individual Property by the expiration or sooner termination of this Agreement shall be treated as a daily tenancy at sufferance as to that Property which Tenant shall be holding over at a daily rate equal to the Holdover Factor multiplied by the Minimum Rent allocable to such Property divided by 365, with such allocation being determined based on the EBITDAR for such Property as a percentage of EBITDAR for all Properties, as reflected in the most recent statements provided by Tenant in accordance with Section 17.2 (if Tenant has provided them on a property-by-property basis), and, otherwise, the Minimum Rent allocable to such Property as reasonably determined by Landlord. The "Holdover Factor" shall be 1.1 for the first thirty (30) days of such failure, and 1.25 for the thirty-first (31st) through sixtieth (60th) days of such failure; and the Holdover Factor shall increase by 0.15 for each subsequent 30-day period of any holding over but the Factor shall never exceed 2 (i.e. shall not exceed 200%). Tenant shall also pay to Landlord all Additional Charges attributable to each such Property during such holding over, and all reasonable out of pocket costs and expenses, if any, actually incurred by Landlord by reason of any such holding over. Otherwise, such holding over shall be on the terms and conditions set forth in this Agreement, to the extent applicable. In addition, if (x) any such holding over shall apply to more than twenty percent (20%) of all of the Properties, rounded to the nearest whole number, or shall exceed six (6) months as to any one or more of the Properties and (y) Landlord shall give Notice to Tenant that it has entered into a lease or other agreement relating to activity at any such Property (either individually or with any other Property) with any Entity which is not an Affiliated Person of Landlord, Tenant shall indemnify Landlord from all loss, cost or liability it shall incur due to any inability of Landlord to deliver possession of any such Property per such lease or other contract due to Tenant's holdover, provided that Tenant shall have the right to control the defense of any claim for which it is providing indemnification. Nothing contained herein shall constitute the consent, express or implied, of Landlord to the holding over of Tenant after the expiration or earlier termination of this Agreement. For the avoidance of doubt, Tenant's exercise of its rights under Section 4.2.3 shall not constitute a holding over.

ARTICLE 14 LANDLORD DEFAULT

If Landlord shall default in the performance or observance of any of its covenants or obligations set forth in this Agreement or any obligation of Landlord, if any, under any agreement affecting the Leased Property, the performance of which is not Tenant's obligation pursuant to this Agreement, and any such default shall continue for a period of thirty (30) days after Notice thereof from Tenant to Landlord and any applicable Property Mortgagee, or such additional period as may be reasonably required to correct the same, Tenant may declare the occurrence of a "Landlord Default" by a second Notice to Landlord and to such Property Mortgagee. Thereafter, Tenant may forthwith cure the same and, subject to the provisions of the following paragraph, invoice Landlord for costs and expenses (including reasonable attorneys' fees and court costs) incurred by Tenant in curing the same, together with interest thereon (to the

extent permitted by law) from the date Landlord receives Tenant's invoice until paid, at the Overdue Rate. Tenant shall have no right to terminate this Agreement for any default by Landlord hereunder and no right, for any such default, to offset or counterclaim against any Rent or other charges due hereunder.

If Landlord shall in good faith dispute the occurrence of any Landlord Default and Landlord, before the expiration of the applicable cure period, shall give Notice thereof to Tenant, setting forth, in reasonable detail, the basis therefor, no Landlord Default shall be deemed to have occurred and Landlord shall have no obligation with respect thereto until final adverse determination thereof. If Tenant and Landlord shall fail, in good faith, to resolve any such dispute within ten (10) days after Landlord's Notice of dispute, either may submit the matter for resolution in accordance with Article 22.

ARTICLE 15 PURCHASE OF TENANT'S PERSONAL PROPERTY

Landlord shall have the option to purchase Tenant's Personal Property and any other tangible personal property of any of Tenant's subtenants which are Affiliated Persons of and Controlled by Tenant which is used in connection with the operation of any Travel Center, at the expiration or sooner termination of this Agreement, for an amount equal to the then fair market value thereof (current replacement cost as determined by agreement of the parties or, in the absence of such agreement, appraisal), subject to, and with appropriate price adjustments for, all liabilities assumed such as equipment leases, conditional sale contracts and other encumbrances securing such liabilities to which such Tenant's Personal Property or property of such subtenant is subject.

ARTICLE 16 SUBLETTING AND ASSIGNMENT

16.1 <u>Subletting and Assignment</u>. Except as provided in <u>Section 16.3</u>, Tenant shall not, without Landlord's prior written consent (which consent shall not be unreasonably withheld or delayed so long, as immediately after giving effect to any such transaction, Landlord will be the beneficiary of a Guaranty from a Person that, immediately following such transfer, is a Qualifying Guarantor), assign, mortgage, pledge, hypothecate, encumber or otherwise transfer this Agreement or sublease or permit the sublease (which term shall be deemed to include the granting of concessions, licenses, sublicenses and the like), of the Leased Property, or any portion thereof, or suffer or permit this Agreement or the leasehold estate created hereby or any other rights arising under this Agreement to be assigned, transferred, mortgaged, pledged, hypothecated or encumbered, in whole or in part, whether voluntarily or involuntarily, or permit the use or operation of the Leased Property, or any portion thereof, by anyone other than Tenant.

If Tenant's interest in this Agreement is assigned, Landlord may collect the rents due hereunder from the assignee. If the Leased Property, or any portion thereof, is sublet (or occupied by anybody other than Tenant), then, upon the occurrence and during the continuance of an Event of Default, Landlord may collect the rents due hereunder from the subtenant or occupant, as the case may be (and, for certainty, all rents so collected shall be credited toward Tenant's payment obligations hereunder). No such collection shall be deemed a waiver of the

Received by OWRD SEP 0 2 2025

provisions set forth in the first paragraph of this Section 16.1, the acceptance by Landlord of such assignee, subtenant or occupant, as the case may be as a true. Section 16.3.2, a release of Tenant from the future performance by Tenant of its covenants. agreements or obligations contained in this Agreement.

Except as expressly set forth herein, no subletting or assignment shall in any way impair the continuing primary liability of Tenant hereunder or any Guarantor, and no consent to any subletting or assignment in a particular instance shall be deemed to be a waiver of the prohibition set forth in this Section 16.1. An original counterpart of each assignment and assumption of this Agreement, duly executed by Tenant and such assignee, shall be delivered to Landlord substantially contemporaneously with execution, and (a) the assignee shall assume in writing and agree to keep and perform all of the terms of this Agreement on the part of Tenant to be kept and performed from and after such assignment and (b) the assignor shall automatically be released from any covenants, agreements and obligations of Tenant hereunder if and to the extent that they were assumed by such assignee. If, in connection with any assignment of Tenant's interest in this Agreement, Tenant delivers a replacement Guaranty to Landlord from a Qualifying Guarantor, as contemplated by the Guaranty, then the then-existing Guarantor automatically shall be released from any obligations under its Guaranty that are obligations of such new Qualifying Guarantor pursuant to such replacement Guaranty or otherwise first arise or accrue after the date of such replacement Guaranty (it being understood that in no circumstance will the then-existing Guarantor be deemed liable for any obligations that first arise or accrue under this Agreement from and after the date of such replacement Guaranty).

No assignment, subletting or occupancy shall affect any Permitted Use. Any subletting, assignment or other transfer of Tenant's interest under this Agreement in contravention of this Article 16 shall be voidable at Landlord's option.

Required Sublease Provisions. Except for subleases which are terminable at will by Tenant on not more than sixty (60) days' prior notice, any sublease of all or any portion of the Leased Property entered into on or after the Effective Date shall provide (a) that the subtenant shall, at Landlord's or Tenant's request pursuant to Tenant's obligations or Landlord's rights under Section 5.3 or Article 15, transfer as so requested any of its Operating Rights and/or other property relating to such Leased Property (and shall be deemed to have granted Landlord the power of attorney with respect to its Operating Rights and other property as Tenant has granted pursuant to the second sentence of the second paragraph of Section 5.3) at the end of such sublease; (b) that it is subject and subordinate to this Agreement and to the matters to which this Agreement is or shall be subject or subordinate; (c) that in the event of termination of this Agreement or reentry or dispossession of Tenant by Landlord under this Agreement, Landlord may, at its option, terminate such sublease or take over all of the right, title and interest of Tenant, as sublessor under such sublease, and such subtenant shall, at Landlord's option, attorn to Landlord pursuant to the then executory provisions of such sublease, except that neither Landlord nor any Property Mortgagee, as holder of a mortgage or as Landlord under this Agreement, if such mortgagee succeeds to that position, shall (i) be liable for any act or omission of Tenant under such sublease, (ii) be subject to any credit, counterclaim, offset or defense which theretofore accrued to such subtenant against Tenant, (iii) be bound by any previous modification of such sublease not consented to in writing by Landlord or by any previous prepayment of more than one (1) month's rent, (iv) be bound by any covenant of Tenant to undertake or complete any

construction of the applicable Property, or any portion thereof, (v) be required to account for any security deposit of the subtenant other than any security deposit actually delivered to Landlord by Tenant, (vi) be bound by any obligation to make any payment to such subtenant or grant any credits, except for services, repairs, maintenance and restoration provided for under the sublease that are performed after the date of such attornment, (vii) be responsible for any monies owing by Tenant to the credit of such subtenant unless actually delivered to Landlord by Tenant, or (viii) be required to remove any Person occupying any portion of the Leased Property; and (d) in the event that such subtenant receives a written Notice from Landlord or any Property Mortgagee stating that this Agreement has terminated, such subtenant shall thereafter be obligated to pay all rentals accruing under such sublease directly to the party giving such Notice or as such party may direct. Such sublease shall provide that the subtenant thereunder shall, at the request of Landlord, execute a suitable instrument in confirmation of such agreement to attorn. An original counterpart of each such sublease, duly executed by Tenant and such subtenant, shall be delivered promptly to Landlord and Tenant shall remain primarily liable, as principal rather than as surety, for the prompt payment of the Rent and for the performance and observance of all of the covenants and conditions to be performed by Tenant hereunder.

The provisions of this <u>Section 16.2</u> shall not be deemed a waiver of the provisions set forth in the first paragraph of <u>Section 16.1</u>.

16.3 Permitted Subleases and Assignments.

16.3.1 Subleases. Subject to the provisions of Section 16.2 and Section 16.4 and any other express conditions or limitations set forth herein, Tenant may, in each instance without consent, (a) enter into third party agreements (including, without limitation, franchise, management, operations or dealer-supply agreements) or sublease space at any Property for fuel or other energy station, restaurant/food service or mechanical repair purposes or other concessions in furtherance of the Permitted Use, so long as such subleases will not violate any Legal Requirements or Insurance Requirements, and Tenant shall provide such additional insurance coverage applicable to the activities to be conducted in such subleased space as Landlord and any Property Mortgagee may reasonably require, and/or (b) enter into one or more subleases or licenses with Affiliated Persons of Tenant with respect to the Leased Property, or any portion thereof, provided such subleases or licenses or sublicenses do not grant any rights with respect to the Leased Property beyond the Term. Landlord and Tenant acknowledge and agree that if Tenant enters into one (1) or more subleases, licenses or sublicenses with Affiliated Persons of Tenant with respect to any Property, or any portion thereof, in accordance with the preceding clause (b), Tenant may allocate the rent and other charges with respect to the affected Property in any reasonable manner; provided, however, that such allocation shall not affect Tenant's (nor Guarantor's) liability for the Rent and other obligations of Tenant under this Agreement.

16.3.2 <u>Tenant Assignments</u>. Tenant may, upon prior Notice to Landlord but without Landlord's consent, assign all (but not less than all) of its right, title and interest in this Agreement to any Affiliated Person of Tenant, provided that (i) such assignment does not violate any Legal Requirements, (ii) either (a) the Guaranty remains in effect irrespective of such assignment and the Guarantor, at the time of such assignment, is a Qualifying Guarantor, or (b) a Qualifying Guarantor delivers a replacement Guaranty to Landlord, as contemplated by the

SEP 0 2 2025
Salem, OR

Guaranty, and (iii) the assignee assumes all obligations of Tenant hereunder. Upon Notice to Landlord from Tenant of such an assignment to an Affiliated Person of Tenant, which Notice shall include a copy of the assignment and assumption agreement, the assignor shall automatically be released from any covenants, agreements and obligations of Tenant hereunder if and to the extent accruing following the date of such assignment.

- 16.3.3 <u>Landlord Assignments</u>. It shall be a condition to the effectiveness of Landlord's assignment of this Agreement that the applicable assignee has assumed in writing and agreed to keep and perform all of the terms of this Agreement on the part of Landlord to be kept and performed from and after such assignment, and Landlord shall promptly deliver the agreement memorializing such assignment and assumption to Tenant.
- 16.4 <u>Sublease Limitation</u>. Anything contained in this Agreement to the contrary notwithstanding, Tenant shall not sublet or sublicense the Leased Property, or any portion thereof, on any basis such that the rental to be paid by any sublessee or sublicensee thereunder would be based, in whole or in part, on the net income or profits derived by the business activities of such sublessee or sublicensee.

ARTICLE 17 ESTOPPEL CERTIFICATES AND OPERATING STATEMENTS

- 17.1 Estoppel Certificates. At any time and from time to time, but not more than a reasonable number of times per year, upon not less than ten (10) Business Days prior Notice by either party, the party receiving such Notice shall furnish to the other an Officer's Certificate certifying that this Agreement is unmodified and in full force and effect (or that this Agreement is in full force and effect as modified and setting forth the modifications), the date to which the Rent has been paid, that no Default or an Event of Default has occurred and is continuing or, if a Default or an Event of Default shall exist, specifying in reasonable detail the nature thereof, and the steps being taken to remedy the same, and such additional information as the requesting party may reasonably request. Any such certificate furnished pursuant to this Article 17 may be relied upon by the requesting party, its lenders and any prospective purchaser or mortgagee of the Leased Property, or any portion thereof, or the leasehold estate created hereby.
- 17.2 Operating Statement. Tenant shall (a) furnish to Landlord, within forty-five (45) days after the end of each fiscal quarter, the following unaudited information: aggregate EBITDA and the corresponding rent, in each case on a combined basis for all Properties and all "Properties" leased pursuant to the Other Leases (as defined therein). In addition: (b) if required of Landlord by an unaffiliated mortgage lender for purposes of underwriting a financing to be secured by the Leased Property (i.e., not on a continuing basis), Tenant will provide unaudited site-level statements that show total gross profit, total site-level operating expenses and 4-wall EBITDAR for the then-current year and each of the two years prior thereto (and Landlord hereby agrees to (and to require its lender to agree, for Tenant's benefit, to) use any such information solely as reasonably required in connection with the underwriting of one or more loans secured in whole or in part by the Leased Properties); and (c) if required by an unaffiliated mortgage lender pursuant to the documents governing the loan, Tenant also will provide, on a quarterly basis, aggregate unaudited EBITDA of the Properties. Together with the furnishing of any such

financial data to Landlord under this Article 17, Tenant shall deliver to Landlord a Financial Officer's Certificate.

ARTICLE 18 LANDLORD'S RIGHT TO INSPECT

Tenant shall permit Landlord and its authorized representatives to inspect the Leased Property, or any portion thereof, during usual business hours upon not less than forty-eight (48) hours' Notice and to make such repairs as Landlord is permitted or required to make pursuant to the terms of this Agreement, provided that any inspection or repair by Landlord or its representatives will not unreasonably interfere with Tenant's use and operation of the Leased Property and further provided that in the event of an emergency, as determined by Landlord in its reasonable discretion, prior Notice shall not be necessary.

ARTICLE 19 EASEMENTS; ZONING

- 19.1 <u>Grant of Easements</u>. Landlord shall not, during the Term, grant, create or otherwise cause to exist any rights-of-way or access rights, easements, Liens or Encumbrances upon the Leased Property without Tenant's prior written consent (not to be unreasonably withheld, conditioned or delayed), but Encumbrances to secure borrowing or other means of financing or refinancing in each case pursuant to and in accordance with <u>Article 20</u> shall not be prohibited. Provided no Event of Default has occurred and is continuing, Landlord will join in granting and, if necessary, modifying or abandoning such rights-of-way, easements and other interests as may be reasonably requested by Tenant for ingress and egress, and electric, telephone, gas, water, sewer and other utilities so long as:
 - (a) the instrument creating, modifying or abandoning any such easement, right-of-way or other interest is satisfactory to and approved by Landlord (which approval shall not be unreasonably withheld, delayed or conditioned); and
 - (b) Landlord receives an Officer's Certificate from Tenant stating (i) that such grant, modification or abandonment is not detrimental to the proper conduct of business on such Property, (ii) the consideration, if any, being paid for such grant, modification or abandonment (which consideration shall be paid by Tenant), (iii) that such grant, modification or abandonment does not impair the use or value of such Property for the Permitted Use, and (iv) that, for as long as this Agreement shall be in effect, Tenant will perform all obligations, if any, of Landlord under any such instrument.
- 19.2 Exercise of Rights by Tenant. So long as no Event of Default has occurred and is continuing, Tenant shall have the right to exercise all rights of Landlord under the Easement Agreements and, in connection therewith, Landlord shall execute and promptly return to Tenant such documents as Tenant shall reasonably request. Tenant shall perform all obligations of Landlord under the Easement Agreements.
- 19.3 <u>Permitted Encumbrances</u>. Any agreements entered into in accordance with this <u>Article 19</u> or in accordance with <u>Article 19</u> during the term of the Prior Lease shall be deemed a <u>Permitted Encumbrance</u>.

19.4 **Zoning**. Landlord shall not, during the Term, initiate or agree to appearing reclassification for the Property or any portion thereof without Tenant's prior written consent (not to be unreasonably withheld, conditioned or delayed).

ARTICLE 20 PROPERTY MORTGAGES

- 20.1 <u>Landlord May Grant Liens</u>. Without the consent of Tenant, Landlord may, from time to time, directly or indirectly, create or otherwise cause to exist any Lien, encumbrance or title retention agreement ("<u>Encumbrance</u>") upon the Leased Property, or any portion thereof, or interest therein, to secure any borrowing or other means of financing or refinancing; provided, that Landlord has complied with the requirements of this Agreement relating thereto; and, provided further, that under no circumstances shall any such borrowing, financing or refinancing or Encumbrance granted by Landlord in connection therewith adversely affect the rights and privileges of Tenant under this Agreement in any material respect or increase in any respect the nature, scope or amount of any obligations or liabilities (including contingent liabilities) of Tenant beyond those set forth in this Agreement, except as provided in Section 20.2.
- 20.2 Subordination of Lease. This Agreement and any and all rights of Tenant hereunder are and shall be subject and subordinate to any ground or master lease, and all renewals, extensions, modifications and replacements thereof, and to all mortgages and deeds of trust, which may now or hereafter affect the Leased Property, or any portion thereof, or any improvements thereon and/or any of such leases, whether or not such mortgages or deeds of trust shall also cover other lands and/or buildings and/or leases, to each and every advance made or hereafter to be made under such mortgages and deeds of trust, and to all renewals, modifications, replacements and extensions of such leases and such mortgages and deeds of trust and all consolidations of such mortgages and deeds of trust, provided that no such subordination shall be required unless Landlord shall comply with its obligations under the last two sentences of this Section 20.2. This section shall be self-operative (in accordance with its terms) and no further instrument of subordination shall be required to give effect hereto. In confirmation of such subordination, Tenant shall promptly execute, acknowledge and deliver any instrument that Landlord, the lessor under any such lease or the holder of any such mortgage or the trustee or beneficiary of any deed of trust or any of their respective successors in interest may reasonably request to evidence such subordination. Any lease to which this Agreement is, at the time referred to, subject and subordinate is herein called "Superior Lease" and the lessor of a Superior Lease or its successor in interest at the time referred to is herein called "Superior Landlord" and any mortgage or deed of trust to which this Agreement is, at the time referred to, subject and subordinate is herein called "Superior Mortgage" and the holder, trustee or beneficiary of a Superior Mortgage is herein called "Superior Mortgagee". Tenant shall have no obligations under any Superior Lease or Superior Mortgage other than those expressly set forth in this Section 20.2.

If any Superior Landlord or Superior Mortgagee or the nominee or designee of any Superior Landlord or Superior Mortgagee shall succeed to the rights of Landlord under this Agreement (any such person, "Successor Landlord"), whether through possession or foreclosure action or delivery of a new lease or deed, or otherwise, at such Successor Landlord's request,

Tenant shall attorn to and recognize the Successor Landlord as Tenant's landlord under this Agreement and Tenant shall promptly execute and deliver any instrument that such Successor Landlord may reasonably request to evidence such attornment (provided that such instrument does not alter the terms of this Agreement), whereupon, this Agreement shall continue in full force and effect as a direct lease between the Successor Landlord and Tenant upon all of the terms, conditions and covenants as are set forth in this Agreement, except that the Successor Landlord (unless formerly the landlord under this Agreement or its nominee or designee) shall not be (a) liable in any way to Tenant for any act or omission, neglect or default on the part of any prior Landlord under this Agreement, (b) responsible for any monies owing by or on deposit with any prior Landlord to the credit of Tenant (except to the extent actually paid or delivered to the Successor Landlord), (c) subject to any counterclaim or setoff which theretofore accrued to Tenant against any prior Landlord, (d) bound by any modification of this Agreement subsequent to such Superior Lease or Superior Mortgage, or by any previous prepayment of Rent for more than one (1) month in advance of the date due hereunder, which was not approved in writing by the Superior Landlord or the Superior Mortgagee thereto, (e) liable to Tenant beyond the Successor Landlord's interest in the Leased Property and the rents, income, receipts, revenues, issues and profits issuing from the Leased Property, or (f) required to remove any Person occupying the Leased Property or any part thereof, except if such Person claims by, through or under the Successor Landlord. Tenant agrees at any time and from time to time to cooperate with Landlord at Landlord's expense in connection with any reasonable request made to facilitate any financing secured by all or any of the Leased Property, and to execute a suitable instrument in confirmation of Tenant's agreement to attorn, as aforesaid, and Landlord agrees to provide Tenant with an instrument of nondisturbance and attornment from each such Superior Mortgagee and Superior Landlord (other than the lessors under any Ground Leases) in form and substance reasonably satisfactory to Tenant. Notwithstanding the foregoing, any Successor Landlord shall be liable to pay to Tenant any portions of insurance proceeds or Awards received by Landlord or the Successor Landlord required to be paid to Tenant pursuant to the terms of this Agreement; and, as a condition to any subordination of this Agreement by Tenant to any mortgage, lien or lease in respect of the Leased Property, or any portion thereof, the mortgagee, lienholder or lessor, as applicable, shall expressly agree, for the benefit of Tenant, to recognize (subject to the provisions of (a)-(f) above) Tenant's leasehold interest under this Agreement upon any foreclosure or other succession to the fee interest of Landlord in the Leased Property, which agreement shall be embodied in an instrument in form reasonably satisfactory to Tenant.

20.3 Notice to Mortgagee and Superior Landlord. Subsequent to the receipt by Tenant of Notice from Landlord as to the identity of any Property Mortgagee or Superior Landlord under a lease with Landlord, as ground lessee, which includes the Leased Property, or any portion thereof, as part of the demised premises and which complies with Section 20.1 (which Notice shall be accompanied by a copy of the applicable mortgage or lease), no Notice from Tenant to Landlord as to a default by Landlord under this Agreement shall be effective with respect to a Property Mortgagee or Superior Landlord unless and until a copy of the same is given to such Property Mortgagee or Superior Landlord at the address set forth in the above described Notice, and the curing of any of Landlord's defaults within the applicable notice and cure periods set forth in Article 14 by such Property Mortgagee or Superior Landlord shall be treated as performance by Landlord.

Received by OWRD SEP 0 2 2025

Salem, OR

ARTICLE 21 ADDITIONAL COVENANTS OF LANDLORD AND TENANT

- 21.1 Prompt Payment of Indebtedness. Tenant shall (a) pay or cause to be paid when due all payments of principal of and premium and interest on Tenant's indebtedness for money borrowed and shall not permit or suffer any such indebtedness to become or remain in default beyond any applicable grace or cure period, (b) pay or cause to be paid when due all lawful claims for labor and rents with respect to the Leased Property, and (c) pay or cause to be paid when due all trade payables, except, in each case, other than that referred to in clause (a), to the extent payment is being contested in good faith by appropriate proceedings in accordance with Article 8 and if Tenant shall have set aside on its books adequate reserves with respect thereto in accordance with GAAP, if appropriate, or unless and until foreclosure, distraint sale or other similar proceedings shall have been commenced.
- 21.2 <u>Maintenance of Accounts and Records</u>. Tenant shall keep true records and books of account of Tenant in which full, true and correct entries will be made of dealings and transactions in relation to the business and affairs of Tenant in accordance with GAAP. Tenant shall apply accounting principles in the preparation of the financial statements of Tenant which, in the judgment of and the opinion of its independent public accountants, are in accordance with GAAP, where applicable, except for changes approved by such independent public accountants.
- 21.3 <u>Notice of Litigation, Etc.</u> Tenant shall give prompt Notice to Landlord of any litigation or any administrative proceeding to which it may hereafter become a party of which Tenant has notice or actual knowledge which involves a potential liability equal to or greater than Five Million Dollars (\$5,000,000) or which is reasonably likely to otherwise result in any material adverse change in the business, operations, property, or condition, financial or otherwise, of Tenant.
- 21.4 <u>Distributions, Payments to Affiliated Persons, Etc.</u> Tenant shall not declare, order, pay or make, directly or indirectly, any Distributions or any payment to any Affiliated Person of Tenant (including payments in the ordinary course of business) or set apart any sum or property therefor, or agree to do so, if, at the time of such proposed action, or immediately after giving effect thereto, any Event of Default for the failure to pay money shall have occurred and be continuing. Otherwise, as long as no such Event of Default shall have occurred and be continuing, Tenant may freely make Distributions and payments to Affiliated Persons; provided, however, that any such payments shall at all times be subordinate to Tenant's obligations under this Agreement.
- 21.5 <u>Merger; Sale of Assets</u>. Except as otherwise permitted in <u>Article 16</u>, without Landlord's prior written consent (which consent may not be unreasonably withheld or delayed), Tenant shall not (i) sell, lease (as lessor or sublessor), transfer or otherwise dispose of, or abandon, all or a majority of its assets (including capital stock or other equity interests) or business to any Person, or (ii) merge into or with any other Entity.

21.6 REIT Qualification.

- 21.6.1 The parties hereto intend that Rent and other amounts paid by Tenant hereunder will qualify as "rents from real property" within the meaning of Section 856(d) of the Code, or any similar or successor provision thereto, and this Agreement shall be interpreted consistent with this intent.
- 21.6.2 Anything contained in this Agreement to the contrary notwithstanding, the parties acknowledge and agree that Landlord, in its sole discretion and at no cost or expense to Tenant (including any applicable taxes), may assign this Agreement or any interest herein to another Person (including without limitation, a "taxable REIT subsidiary" (within the meaning of Section 856(1) of the Code)) in order to maintain Landlord's (or any of its Affiliated Persons') status as a "real estate investment trust" (within the meaning of Section 856(a) of the Code); provided, however, Landlord shall be required to comply with any applicable legal requirements related to such transfer; and provided, further, that any such assignment shall be subject to all of the rights of Tenant hereunder.
- 21.6.3 Anything contained in this Agreement to the contrary notwithstanding, upon reasonable request of Landlord, Tenant shall cooperate with Landlord in good faith, and provide such documentation and/or information as may be in Tenant's possession or under Tenant's Control and otherwise readily available to Tenant as shall be reasonably requested by Landlord in connection with verification of Landlord's "real estate investment trust" (within the meaning of Section 856(a) of the Code) compliance requirements; provided that such cooperation and provision of documentation and/or information by Tenant shall not result in any unreimbursed cost, expense or other adverse consequences to Tenant.
- 21.6.4 This <u>Section 21.10</u> is intended to benefit and be enforceable by Landlord and its Affiliated Persons.

ARTICLE 22 ARBITRATION

- 22.1 <u>Disputes</u>. Each party agrees that any disputes, claims or controversies between or among the parties, arising out of or relating to this Agreement (including any such dispute, claim or controversy involving either party's respective shareholders, members, direct and indirect parents, trustees, directors, officers, managers (including, in the case of Landlord, The RMR Group Inc. and The RMR Group LLC), agents or employees and their respective successors and assigns as parties thereto), including disputes, claims or controversies relating to the meaning, interpretation, effect, validity, performance or enforcement of this Agreement, including this arbitration provision (all of which are referred to as "<u>Disputes</u>"), or relating in any way to such a Dispute or Disputes shall, on the demand of any party to the Dispute be resolved through binding and final arbitration in accordance with the Commercial Arbitration Rules (the "<u>Rules</u>") of the American Arbitration Association ("<u>AAA</u>") then in effect, except as those Rules may be modified in this <u>Article 22</u>. For the avoidance of doubt, a Dispute shall include a Dispute made derivatively on behalf of one party against another party.
- 22.2 <u>Selection of Arbitrators</u>. There shall be three (3) arbitrators. If there are only two (2) parties to the Dispute, each party shall select one arbitrator within fifteen (15) days after receipt of a demand for arbitration. Such arbitrators may be affiliated or interested persons of

such parties. If there are more than two (2) parties to the Dispute, all claimants, on the one hand, and all respondents, on the other hand, shall each select, by the vote of a majority of the claimants or the respondents, as the case may be, one arbitrator within fifteen (15) days after receipt of a demand for arbitration. Such arbitrators may be affiliated or interested persons of the claimants or the respondents, as the case may be. If either a claimant (or all claimants) or a respondent (or all respondents) fail to timely select an arbitrator then the party (or parties) who has selected an arbitrator may request the AAA to provide a list of three (3) proposed arbitrators in accordance with the Rules (each of whom shall be neutral, impartial and unaffiliated with any party) and the party (or parties) that failed to timely appoint an arbitrator shall have ten (10) days from the date the AAA provides such list to select one of the three (3) arbitrators proposed by AAA. If such party (or parties) fail to select such arbitrator by such time, the AAA shall select, within fifteen (15) days thereafter, one of the three (3) arbitrators it had proposed as the second arbitrator. The two (2) arbitrators so appointed shall jointly appoint the third and presiding arbitrator (who shall be neutral, impartial and unaffiliated with any party) within fifteen (15) days of the appointment of the second arbitrator. If the third arbitrator has not been appointed within the time limit specified herein, then the AAA shall provide a list of proposed arbitrators in accordance with the Rules, and the arbitrator shall be appointed by the AAA in accordance with a listing, striking and ranking procedure, with each party having a limited number of strikes, excluding strikes for cause.

- 22.3 <u>Location of Arbitration</u>. The place of arbitration shall be Boston, Massachusetts unless otherwise agreed by the parties.
- 22.4 <u>Scope of Discovery</u>. There shall be only limited documentary discovery of documents directly related to the issues in dispute, as may be ordered by the arbitrators. For the avoidance of doubt, it is intended that there shall be no depositions and no other discovery other than limited documentary discovery as described in the preceding sentence.
- Award"), the arbitrators shall be required to follow the Applicable Law set forth in Section 24.13. Any arbitration proceedings or Arbitration Award rendered hereunder and the validity, effect and interpretation of this arbitration provision shall be governed by the Federal Arbitration Act, 9 U.S.C. §1 et seq. The Arbitration Award shall be in writing and may, but shall not be required to, briefly state the findings of fact and conclusions of law on which it is based. Any monetary Arbitration Award shall be made and payable in U.S. dollars free of any tax, deduction or offset. Subject to Section 22.7, each party against which an Arbitration Award assesses a monetary obligation shall pay that obligation on or before the thirtieth (30th) day following the date of such Arbitration Award or such other date as such Arbitration Award may provide.
- Appeals. Notwithstanding any language to the contrary in this Agreement, any Arbitration Award, including but not limited to any interim Arbitration Award, may be appealed pursuant to the AAA's Optional Appellate Arbitration Rules ("Appellate Rules"). An Arbitration Award shall not be considered final until after the time for filing the notice of appeal pursuant to the Appellate Rules has expired. Appeals must be initiated within thirty (30) days of receipt of an Arbitration Award by filing a notice of appeal with any AAA office. Following the appeal process, the decision rendered by the appeal tribunal may be entered in any court having jurisdiction thereof. For the avoidance of doubt, and despite any contrary provision of the

Appellate Rules, the above paragraph relating to costs and expenses shall apply to any appeal pursuant to this Article 22.

- 22.7 <u>Final Judgment</u>. Following the expiration of the time for filing the notice of appeal, or the conclusion of the appeal process set forth in <u>Section 22.6</u>, an Arbitration Award shall be final and binding upon the parties thereto and shall be the sole and exclusive remedy between those parties relating to the Dispute, including any claims, counterclaims, issues or accounting presented to the arbitrators. Judgment upon an Arbitration Award may be entered in any court having jurisdiction. To the fullest extent permitted by law, no application or appeal to any court of competent jurisdiction may be made in connection with any question of law arising in the course of arbitration or with respect to any Arbitration Award made, except for actions relating to enforcement of this <u>Article 22</u> or any arbitral award issued hereunder, and except for actions seeking interim or other provisional relief in aid of arbitration proceedings in any court of competent jurisdiction.
- 22.8 <u>Intended Beneficiaries</u>. This Article 22 is intended to benefit and be enforceable by the parties and their respective successors and assigns.

ARTICLE 23 REPRESENTATIONS

- 23.1 <u>Landlord's Representations</u>. Landlord represents and warrants to Tenant as of the Effective Date as follows:
- 23.1.1 Landlord owns fee simple title to the Leased Property other than Properties which are subject to Ground Leases, and Landlord is the sole lessee under the Ground Leases.
- 23.1.2 Landlord is duly organized, validly existing and in good standing under the laws of its state of formation, is qualified to do business and in good standing in the State in which each Property is located (to the extent Landlord is required to be so by applicable Legal Requirements) and has full power, authority and legal right to execute and deliver and to perform and observe the provisions of this Agreement to be observed and/or performed by Landlord.
- 23.1.3 This Agreement has been duly authorized, executed and delivered by Landlord and constitutes and will constitute the valid and binding obligations of Landlord enforceable against Landlord in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, creditors' rights laws and/or general principles of equity.
- 23.1.4 The execution and delivery of this Agreement and compliance with the provisions hereof will not result in (i) a material breach or violation of (A) any Legal Requirements applicable to Landlord now in effect, (B) the organizational or charter documents of Landlord, (C) any judgment, order or decree of any Government Agency binding upon Landlord or (D) any material agreement or instrument to which Landlord is a counterparty or by which it is bound or (ii) the acceleration of any material obligation of Landlord.
- 23.2 <u>Tenant's Representations</u>. Tenant represents and warrants to Landlord as of the Effective Date as follows:

- 23.2.1 Tenant is duly organized, validly existing and in good standing under the laws of its state of formation, is qualified to do business and in good in the State in which each Property is located (to the extent Tenant is required to be so by applicable Legal Requirements) and has full power, authority and legal right to execute and deliver and to perform and observe the provisions of this Agreement to be observed and/or performed by Tenant.
- 23.2.2 This Agreement has been duly authorized, executed and delivered by Tenant, and constitutes and will constitute the valid and binding obligations of Tenant enforceable against Tenant in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, creditors' rights laws and/or general principles of equity.
- 23.2.3 The execution and delivery of this Agreement and compliance with the provisions hereof will not result in (i) a material breach or violation of (A) any Legal Requirements applicable to Tenant now in effect, (B) the organizational or charter documents of Tenant, (C) any judgment, order or decree of any Government Agency binding upon Tenant or (D) any material agreement or instrument to which Tenant is a counterparty or by which it is bound or (ii) the acceleration of any material obligation of Tenant.

ARTICLE 24 MISCELLANEOUS

- Limitation on Payment of Rent. All agreements between Landlord and Tenant herein are hereby expressly limited so that in no contingency or event whatsoever, whether by reason of acceleration of Rent, or otherwise, shall the Rent or any other amounts payable to Landlord under this Agreement exceed the maximum permissible under applicable law, the benefit of which may be asserted by Tenant as a defense, and if, from any circumstance whatsoever, fulfillment of any provision of this Agreement, at the time performance of such provision shall be due, shall involve transcending the limit of validity prescribed by law, or if from any circumstances Landlord should ever receive as fulfillment of such provision such an excessive amount, then, <u>ipso facto</u>, the amount which would be excessive shall be applied to the reduction of the installment(s) of Minimum Rent next due and not to the payment of such excessive amount. This provision shall control every other provision of this Agreement and any other agreements between Landlord and Tenant.
- 24.2 <u>No Waiver</u>. No failure by Landlord or Tenant to insist upon the strict performance of any term hereof or to exercise any right, power or remedy consequent upon a breach thereof, and no acceptance of full or partial payment of Rent during the continuance of any such breach, shall constitute a waiver of any such breach or of any such term. To the maximum extent permitted by law, no waiver of any breach shall affect or alter this Agreement, which shall continue in full force and effect with respect to any other then existing or subsequent breach.
- 24.3 <u>Remedies Cumulative</u>. To the maximum extent permitted by law, each legal, equitable or contractual right, power and remedy of Landlord or Tenant, now or hereafter provided either in this Agreement or by statute or otherwise, shall be cumulative and concurrent and shall be in addition to every other right, power and remedy and the exercise or beginning of the exercise by Landlord or Tenant (as applicable) of any one or more of such rights, powers and

remedies shall not preclude the simultaneous or subsequent exercise by Landlord of any or all of such other rights, powers and remedies.

- 24.4 <u>Severability</u>. Any clause, sentence, paragraph, section or provision of this Agreement held by a court or arbitral tribunal of competent jurisdiction to be invalid, illegal or ineffective shall not impair, invalidate or nullify the remainder of this Agreement, but rather the effect thereof shall be confined to the clause, sentence, paragraph, section or provision so held to be invalid, illegal or ineffective, and this Agreement shall be construed as if such invalid, illegal or ineffective provisions had never been contained therein.
- 24.5 <u>Acceptance of Surrender</u>. No surrender to Landlord of this Agreement or of the Leased Property or any part thereof, or of any interest therein, shall be valid or effective unless agreed to and accepted in writing by Landlord and no act by Landlord or any representative or agent of Landlord, other than such a written acceptance by Landlord, shall constitute an acceptance of any such surrender.
- 24.6 <u>No Merger of Title</u>. It is expressly acknowledged and agreed that it is the intent of the parties that there shall be no merger of this Agreement or of the leasehold estate created hereby by reason of the fact that the same Person may acquire, own or hold, directly or indirectly, this Agreement or the leasehold estate created hereby and the fee estate or ground landlord's interest in the Leased Property.
- 24.7 <u>Conveyance by Landlord</u>. If Landlord or any successor owner of all or any portion of the Leased Property shall convey all or any portion of the Leased Property in accordance with the terms hereof other than as security for a debt, and the grantee or transferee of such of the Leased Property shall expressly assume all obligations of Landlord hereunder arising or accruing from and after the date of such conveyance or transfer, Landlord or such successor owner, as the case may be, shall thereupon be released from all future liabilities and obligations of Landlord under this Agreement with respect to such of the Leased Property arising or accruing from and after the date of such conveyance or other transfer and all such future liabilities and obligations shall thereupon be binding upon the new owner.
- 24.8 **Quiet Enjoyment**. Landlord covenants and agrees that Tenant shall have the right to peaceably and quietly have, hold and enjoy the Leased Property for the Term, free of hindrance or molestation by Landlord or anyone claiming by, through or under Landlord, so long as no Event of Default is continuing.
 - 24.9 No Recordation. Neither Landlord nor Tenant shall record this Agreement.

24.10 Notices.

(a) Any and all notices, demands, consents, approvals, offers, elections and other communications required or permitted under this Agreement shall be deemed adequately given if in writing and the same shall be delivered either in hand or by mail or Federal Express or similar expedited commercial carrier, addressed to the recipient of the notice, postpaid and registered or certified with return receipt requested (if by mail), or with all freight charges prepaid (if by Federal Express or similar carrier).

(b) All notices required or permitted to be sent hereunder shall be deemed to have been given for all purposes of this Agreement upon the date of receipt or refusal, except that whenever under this Agreement a notice is either received on a day which is not a Business Day or is required to be delivered on or before a specific day which is not a Business Day, the day of receipt or required delivery shall automatically be extended to the next Business Day.

(c) All such notices shall be addressed,

if to Landlord, to:

c/o Service Properties Trust

Two Newton Place

255 Washington Street, Suite 300 Newton, Massachusetts 02458

Attn: President and Chief Financial Officer

Email: thargreaves@rmrgroup.com and bdonley@rmrgroup.com

if to Tenant, to:

c/o TravelCenters of America Inc.

24601 Center Ridge Road Westlake, Ohio 44145

Attn: President and Chief Financial Officer

Email: Gregory.franks@bp.com and babu.rajalingam@bp.com

with a copy to:

BP Products North America Inc. 30 South Wacker Drive, Suite 900

Chicago, Illinois 60606

Attention: Retail Real Estate Manager

Email: daniel.fiden@bp.com

with a copy to:

BP America Inc.

30 South Wacker Drive, Suite 900

Chicago, Illinois 60606

Attention.: Real Estate Attorney Email: william.lockhart@bp.com

- (d) By notice given as herein provided, the parties hereto and their respective successors and assigns shall have the right from time to time and at any time during the term of this Agreement to change their respective addresses effective upon receipt by the other parties of such notice and each shall have the right to specify as its address any other address within the United States of America.
- 24.11 <u>Construction</u>. The recitals to this Agreement are incorporated herein in their entirety as if fully set forth in this Agreement and shall constitute a part of this Agreement. Anything contained in this Agreement to the contrary notwithstanding, all claims against, and liabilities of, Tenant or Landlord arising prior to any date of termination or expiration of this Agreement with respect to the Leased Property shall survive such termination or expiration, and

in no event shall Landlord or Tenant be liable for any consequential or punitive damages suffered by the other party as the result of a breach of this Agreement or otherwise. Neither this Agreement nor any provision hereof may be changed, waived, discharged or terminated except by an instrument in writing signed by the party to be charged. All the terms and provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns. Each term or provision of this Agreement to be performed by Tenant shall be construed as an independent covenant and condition. Time is of the essence with respect to the provisions of this Agreement. Except as otherwise set forth in this Agreement, any obligations of Tenant (including without limitation, any monetary, repair and indemnification obligations) and Landlord shall survive the expiration or sooner termination of this Agreement. Tenant hereby acknowledges that the agreement between Landlord and Tenant to treat this Agreement as a single lease in all respects was and is of primary importance, and a material inducement, to Landlord to enter into this Agreement. Without limiting the generality of the foregoing, the parties hereto acknowledge that this Agreement constitutes a single lease of the Leased Property and is not divisible notwithstanding any references herein to any individual Property and notwithstanding the possibility that certain individual Properties may be deleted herefrom pursuant to the express provisions of this Agreement.

- 24.12 <u>Counterparts</u>; <u>Headings</u>. This Agreement may be executed in two or more counterparts, each of which shall constitute an original, but which, when taken together, shall constitute but one instrument and shall become effective as of the Effective Date. Headings in this Agreement are for purposes of reference only and shall not limit or affect the meaning of the provisions hereof.
- Applicable Law, Etc Except as to matters regarding the internal affairs of Landlord and issues of or limitations on any personal liability of the shareholders and trustees or directors of Landlord for obligations of Landlord, as to which the laws of the State of Maryland shall govern, this Agreement shall be interpreted, construed, applied and enforced in accordance with the laws of The Commonwealth of Massachusetts applicable to contracts between residents of Massachusetts which are to be performed entirely within Massachusetts, regardless of (i) where this Agreement is executed or delivered; or (ii) where any payment or other performance required by this Agreement occurs, or any cause of action otherwise accrues; or (iv) where any action or other proceeding is instituted or pending; or (v) the nationality, citizenship, domicile, principal place of business, or jurisdiction of organization or domestication of any party; or (vi) whether the laws of the forum jurisdiction otherwise would apply the laws of a jurisdiction other than Massachusetts; or (vii) any combination of the foregoing. Notwithstanding the foregoing, the laws of the State shall apply to the perfection and priority of liens upon and the disposition of any Property.
- 24.14 <u>Right to Make Agreement</u>. Each party warrants, with respect to itself, that neither the execution of this Agreement, nor the consummation of any transaction contemplated hereby, shall violate any provision of any law, or any judgment, writ, injunction, order or decree of any court or governmental authority having jurisdiction over it; nor result in or constitute a breach or default under any indenture, contract, other commitment or restriction to which it is a party or by which it is bound; nor require any consent, vote or approval which has not been given or taken, or at the time of the transaction involved shall not have been given or taken. Each party

SEP 0 2 2025
Salem, OR

covenants that it has and will continue to have throughout the term of this Agreement and any extensions thereof, the full right to enter into this Agreement and perform its obligations hereunder.

- 24.15 Confidentiality. Except as may be required by Applicable Law (including, for the avoidance of doubt, the disclosure requirements of applicable securities law) and any quarterly aggregate reporting provided pursuant to clause (a) of Section 17.2, Landlord hereby agrees that Landlord will keep confidential any non-public financial, operational or other information relating to Tenant, Guarantor, the Leased Property or any business conducted thereon and disclosed or made available to Landlord or SVC (or any Person acting on either of their behalf) pursuant to or in connection with this Agreement or any Guaranty, and not to disclose any such information to any Person without the prior written consent of Tenant; provided, however, that either party may, without consent, disclose any such information to such party's Affiliated Persons or to such party's or its Affiliated Person's investors, accountants, attorneys, employees, agents or lenders (each, a "Related Person") to the extent reasonably necessary (a) for such party's business purposes, so long as the recipient of such information shall be required to maintain the confidentiality of such information in the same manner and to the extent as the parties hereunder, or (b) in connection with a dispute undertaken pursuant to Article 22 hereof, it being understood and agreed, in any case, that each party shall be liable to the other parties for any failure by the Related Persons of such first-mentioned party to handle such disclosed information in accordance with this Section 24.15.
- 24.16 <u>Costs</u>; <u>Attorneys</u>' <u>Fees</u>. To the maximum extent permitted by Applicable Law, if any lawsuit or arbitration or other legal proceeding arises in connection with the interpretation or enforcement of this Agreement, the prevailing party therein shall be entitled to receive from the other party the prevailing party's costs and expenses, including reasonable attorneys' fees and the cost and expenses of both parties' arbitrators incurred in connection therewith, in preparation therefor and on appeal therefrom, which amounts shall be included in any decision or judgment therein.
- 24.17 Exculpation. Notwithstanding anything to the contrary contained herein, no recourse under or upon any obligation, representation, warranty, promise or other matter whatsoever under this Agreement shall be had against any of the direct or indirect constituent members, affiliates, partners, shareholders, officers, directors, employees, agents or representatives (collectively, the "Non-Recourse Parties") of Tenant or such Non-Recourse Parties (other than Guarantor under any Guaranty), and Landlord expressly waives and releases, on behalf of itself and its respective successors and assigns, all right to assert any liability whatsoever under or with respect to this Agreement against, or to satisfy any claim or obligation arising hereunder against, any of such Non-Recourse Parties of Tenant or such Non-Recourse Parties or out of any of their assets (other than Guarantor under any Guaranty).
- 24.18 Nonliability of Trustees. THE DECLARATION OF TRUST ESTABLISHING HPT TA PROPERTIES TRUST, A COPY OF WHICH, TOGETHER WITH ALL AMENDMENTS THERETO (THE "DECLARATION"), IS DULY FILED WITH THE DEPARTMENT OF ASSESSMENTS AND TAXATION OF THE STATE OF MARYLAND, PROVIDES THAT THE NAME OF SUCH ENTITY REFERS TO THE TRUSTEES UNDER SUCH DECLARATION COLLECTIVELY AS TRUSTEES, BUT NOT INDIVIDUALLY OR

PERSONALLY, AND THAT NO TRUSTEE, OFFICER, SHAREHOLDER, EMPLOYEE OR AGENT OF SUCH ENTITY SHALL BE HELD TO ANY PERSONAL LIABILITY, JOINTLY OR SEVERALLY, FOR ANY OBLIGATION OF, OR CLAIM AGAINST, SUCH ENTITY. ALL PERSONS DEALING WITH SUCH ENTITY, IN ANY WAY, SHALL LOOK ONLY TO THE ASSETS OF SUCH ENTITY FOR THE PAYMENT OF ANY SUM OR THE PERFORMANCE OF ANY OBLIGATION.

24.19 <u>True Lease and Operating Lease</u>. It is the intent of Landlord and Tenant, and the parties agree, that this Agreement, for federal income tax purposes, is a true lease and that this Agreement does not represent a financing agreement. It is the further intent of Landlord and Tenant, and the parties agree, that this Agreement, for accounting purposes of the Tenant, is an operating lease. Each party shall reflect the transaction represented hereby in all applicable books, records and reports (including income tax filings) of such party in a manner consistent with "true lease" treatment rather than "financing" treatment.

[Remainder of Page Left Blank Intentionally]

Received by OWRD
SEP 0 % 2025

	TA	Salem, OR	
Exhibit	Site No.	Property Address	
A-21	52	100 North Broadway	
		Oak Grove, MO 64075	
A-22	90	103 Prospectors Drive	
		Ogallala, NE 69153	
A-23	362	4700 S. Lincoln Avenue	
		York, NE 68467	
A-24	48	975 State Route 173	
		Bloomsbury, NJ 08804	
A-25	23	HC 69 - Box 120	
		Santa Rosa, NM 88435.	
A-26	2	1101 NC Highway 61	
		Whitsett (Greensboro), NC 27377	
A-27	39	10679 Lancaster Road	
		Hebron, OH 43025	
A-28	29	5551 State Route 193	
		Kingsville, OH 44048	
A-29	59	501 South Morgan Road	
		Oklahoma City (West), OK 73128	
A-30	56	21856 Bents Road, NE	
		Aurora (Portland), OR 97002	
A-31	215	4050 Depot Road	
		Erie (Harborcreek), PA 16510	
A-32	12	7848 Linglestown Road	
		Harrisburg, PA 17112	
A-33	255	289 Howard Baker Highway	
		Pioneer, TN 37847	
A-34	17	6800 Thompson Road	
		Baytown, TX 77522	
A-35	230	704 West Interstate 20	
		Big Spring, TX 79720	
A-36	1	100 N. Carter Road	
		Ashland (Richmond), VA 23005	

Exhibit A-30

TA Site 56 21856 Bents Road, NE Aurora (Portland), OR 97002 SEP 0 2 2025
Salem, OR

A parcel of land located in the Amable Arqualt Donation Land Claim No. 45 in the Southeast one-quarter of Section 9, Township 4 South, Range 1 West of the Williamette Meridian, Marion County, Oregon and more particularly described as follows:

Beginning at an iron rod on the Easterly right-of-way line of County Road No. 425, said point being South 08°07'00" East along the centerline of said County Road a distance of 1,666.58 feet (the deed calls this 1,657.50 feet) and North 73°21'00" East a distance of 30.34 feet from the Southeast corner of the David Crawford Donation Land Claim No. 41; thence continuing North 73°21'00" East a distance of 629.64 feet (the deed calls this 630.05 feet); thence North 30°08'00" East a distance of 721.88 feet (the deed calls this 722.11 feet) to a point on the Southerly line of that parcel of land conveyed to Louis N. Racette et al, and recorded in Volume269, page 508, Marion County Deed Records; thence North 73°21'00" East along the Southerly line of said Racette property a distance of 479.69 feet (the deed calls this 479.64 feet) to a point on the Westerly right-of-way line of Pacific Highway (Interstate No. 5); thence South 30°08'00" West along said Westerly right-of-way line a distance of 1897.70 feet (the deed calls this 1898.17 feet); thence South 42°36'38" West, (the deed calls this South 42°36' West) along said Westerly right-of-way line a distance of 189.54 feet (the deed calls this 189.59 feet); thence South 46°20'55" West (the deed calls this South 46°19'00" West) along said Westerly right-of-way line a distance of 161.22 feet (the deed calls this 161.43 feet); thence North 59°54'00" West along said right-of-way line a distance of 104.21 feet (the deed calls this 104.28 feet); thence continuing along said right-of-way line Northwesterly along a 246.48 foot radius curve to the right, through a central angle of 00°40'02" an arc distance of 2.87 feet (the long chord of said curve bears North 08°29'44" West (the deed calls this North 08°28' West a distance of 2.87 feet); theree South 81°53'00" West along said right-of-way line a distance of 3.65 feet; thence North 59°54'00" West along said right-of-way line a distance of 7.17 feet to a point on the Easterly right-of-way line of County Road No. 425; thence North 08°07'00" West along said Easterly right-of-way line a distance of 900.66 feet (the deed calls this 900.92 feet) to the point of beginning.

Colleen Giltz

From:

Bob Gertzen

Sent:

Thursday, August 28, 2025 4:30 PM

To: Cc: Colleen Giltz Brandi Babusch

Subject:

FW: TA Travel Center Groundwater Application

Hi Colleen,

Will you please print the email below and the check I have ready in the queue, and mail it to the address that is on the check on Friday?

Brandi, please upload an invoice for reimbursement ASAP. No markup.

Thank you,

Bob Gertzen, President/CEO

Utility Services

Phone: (219) 759-0193, ext. 1004

Cell: (219) 309-6546

Address: 10525 US Hwy 30 Bldg #12 Wanatah, IN 46390

Web: www.utilityservicescorp.com

The information transmitted in this e-mail message and attachments is intended only for the use of the individual or entity named above. Distribution to, or review by, unauthorized persons is prohibited. If you have received this transmission in error, immediately notify us by reply email or contact us at (219) 759-0193 and permanently delete this transmission including attachments, if any. Thank you.

From: BENHAM Sarah L * WRD < Sarah.L.BENHAM@water.oregon.gov >

Sent: Tuesday, August 26, 2025 2:04 PM

To: Shay Olsen < solsen@hbh-consulting.com >; LOVRIEN Corie L * WRD < corie.l.lovrien@water.oregon.gov >; Andrey

Chernishov <achernishov@hbh-consulting.com>; WRD_DL_customerservice

<WRD DL customerservice@water.oregon.gov>

Subject: Re: TA Travel Center Groundwater Application

Hello,

OWRD has received and accepted the Groundwater e-application for <u>TA Operating</u> <u>LLC</u>. The temporary application number is: <u>SB-07.</u>

Please print this email and include a copy of it when mailing in the payment. The fees due for this application are: \$2,970.00 (If opting to include recording fee: \$3,885.00)

A copy of the fee calculator is included for your reference:

Received

SEP 0 2 2025

OWRD



Today's Date: Tuesday, August 26, 2025

Dana Apullandan Fan		60 055 00
Base Application Fee.		\$2,355.00
Number of proposed cubic feet per second (cfs) to be appropriated. (1 cfs = 448.83 gallons per minute)	0.18	\$615.00
Number of proposed Use's for the appropriated water. (i.e. Irrigation, Supplemental Irrigation, Pond Maintenance, Industrial, Commercial, etc) *		
Number of proposed groundwater points of appropriation. (i.e. number of wells) (include all injection wells, if applicable) **		
Subtota		
Permit Recording Fee. ***		\$915.00
* the 1st Water Use is included in the base cost. ** the 1st groundwater point of appropriation is included in the base cost. *** the Permit Recording Fee is not required when the application is submitted but, must be paid before a permit will be issued. It is fully refundable if a permit is not issued. If the recording fee is not paid prior to issuance of the Final Order, permit issuance will be delayed.		
Estimated cost of Permit Application		\$3,885.00

Thanks,

Sarah Benham Customer Service Representative Water Right Services Division 725 Summer Street NE Ste A Salem, Oregon 97301 (503) 986-0810



Integrity | Service | Technical Excellence | Teamwork | Forward-Looking

NOTE: The Salem office is now open to the public. Given that many staff will continue teleworking remotely or have job duties that take them into the field on a regular basis, availability of staff in the office is not guaranteed 8 a.m. - 5 p.m. every day. The Salem office of OWRD is closed for customer service drop-ins from Noon – 1pm. **Customers and visitors are encouraged to schedule an appointment in advance if they wish to meet in person with specific staff members.** Alternative methods for meeting, such as by phone or virtually via Teams, are also available.



SEP 0 2 2025

